

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Löschel, Andreas

## Working Paper Technological change in economic models of environmental policy: A survey

Nota di Lavoro, No. 4.2002

**Provided in Cooperation with:** Fondazione Eni Enrico Mattei (FEEM)

*Suggested Citation:* Löschel, Andreas (2002) : Technological change in economic models of environmental policy: A survey, Nota di Lavoro, No. 4.2002, Fondazione Eni Enrico Mattei (FEEM), Milano

This Version is available at: https://hdl.handle.net/10419/119613

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU



Fondazione Eni Enrico Mattei

## Technological Change in Economic Models of Environmental Policy: A Survey

Andreas Löschel\*

NOTA DI LAVORO 4.2002

## JANUARY 2002

CLIM - Climate Change Modelling and Policy

\*Centre for European Economic Research, ZEW and University of Mannheim

This paper can be downloaded without charge at:

The Fondazione Eni Enrico Mattei Note di Lavoro Series Index: http://www.feem.it/web/activ/\_activ.html

Social Science Research Network Electronic Paper Collection: http://papers.ssrn.com/abstract=XXXXXX

Fondazione Eni Enrico Mattei Corso Magenta, 63, 20123 Milano, tel. +39/02/52036934 – fax +39/02/52036946 E-mail: letter@feem.it C.F. 97080600154

### SUMMARY

This paper provides an overview of the treatment of technological change in economic models of environmental policy. Numerous economic modeling studies have confirmed the sensitivity of mid- and long-run climate change mitigation cost and benefit projections to assumptions about technology costs. In general, technical progress is considered to be a non-economic, exogenous variable in global climate change modeling. However, there is overwhelming evidence that technological change is not an exogenous variable but to an important degree endogenous, induced by needs and pressures. Hence, some environment-economy models treat technological change as endogenous, responding to socioeconomic variables. Three main elements in models of technological innovation are: (i) corporate investment in research and development, (ii) spillovers from R&D, and (iii) technology learning, especially learningby-doing. The incorporation of induced technological change in different types of environmental-economic models tends to reduce the costs of environmental policy, accelerates abatement and may lead to positive spillover and negative leakage.

**Keywords**: Climate policy, exogenous technological change, induced technological change, environment-economy models

JEL: C50, C68, O30, Q25

## NON TECHNICAL ABSTRACT

The threat of climate change, potentially produced by the growing accumulation of greenhouse gases (GHG) in the atmosphere, has led to an increasing number of empirical models for climate change policy analysis. Numerous modeling studies have shown the sensitivity of midand long-run climate change mitigation cost and benefit projections to assumptions about technology. Technological change is in general considered to be a non-economic, exogenous variable in energy-economy models. Economic activities and policies have then no impact on research, development, and diffusion of new technologies. However, there is overwhelming evidence that technological change is not an exogenous variable, but to an important degree endogenous, induced by needs and pressures. Hence, a new generation of environmental-economic models treats technological change as endogenous, i.e. responding to socioeconomic (policy) variables, e.g. prices, investment in R&D, or cumulative production.

The aim of this paper is to provide an overview of how exogenous and endogenous technological change is represented in applied environmenteconomy models. First, a taxonomy of different model types (bottom-up, top-down, integrated assessment models) is given. Then, exogenous specifications of technical change such as the autonomous energy efficiency parameter, the specification of backstop technologies and technology snapshots are explored. Even though the theory of induced technological change is still in development at present, three main approaches to incorporate induced technical progress can be identified: (i) corporate investment in research and development, (ii) spillovers from R&D and (iii) technology learning, especially learning-by-doing. It is shown how technical change is accounted for in well-known economic models of climate change. The incorporation of induced technological change tends to reduce the costs of environmental policy, accelerates abatement and may lead to positive spillover and negative leakage.

Future extensions of the presented approaches should be targeted towards a further endogenization and an improved realism in the modeling of the process of technological change, e. g., the incorporation of path-dependence and inertia, the discontinuity in the process of technological change, and the heterogeneity in firm behavior and investment incentives. In addition, the need for calibration of R&D related model parameters requests extended econometric studies to provide an empirical background for modeling.

## CONTENTS

1.	Introduction	3
2.	Modeling approaches: bottom-up vs. top-	5
	down	
3.	Exogenous specification of technological	8
	change	
4.	Endogenous specification of technological	11
	change	
5.	Economic models of environmental policy	21
6.	Implications of exogenous and induced	26
	technical change	
7.	Conclusion	30
	References	32

## TECHNOLOGICAL CHANGE IN ECONOMIC MODELS OF ENVIRONMENTAL POLICY: A Survey

#### 1. Introduction

The threat of climate change, potentially produced by the growing accumulation of greenhouse gases (GHG) in the atmosphere, has led to an increasing number of empirical models for climate change policy analysis. Numerous modeling studies have shown the sensitivity of mid- and long-run climate change mitigation cost and benefit projections to assumptions about technology (EMF 1996). Technological change (TC), that is increases in outputs without increases in productive inputs, can lower the cost of GHG abatement policies through product innovations, i.e. higher energy-efficiency of existing and new products, and process innovations, i.e. higher energy efficiency of manufacturing processes, cost reductions in low-emission energy conversion and improvements in fossil energy conversion. Technological change is in general considered to be a non-economic, exogenous variable in energy-economy models. Economic activities and policies have then no impact on research, development, and diffusion of new technologies. The emphasis is placed upon showing the mere effect of technical change, but not on how technology development occurs. However, there is overwhelming evidence that technological change is not an exogenous variable, that can be simply defined outside the model, but to an important degree endogenous, induced by needs and pressures (Grubb et al. 1995). Especially over the longer time horizon typical for climate policy analysis, models incorporating induced technological change may project total costs of abatement that are substantially lower than those reported by conventional models with exogenous technical change. Hence, a new generation of environmental-economic models treats technological change as endogenous, i.e. responding to socio-economic (policy) variables, e.g. prices, investment in R&D, or cumulative production. This approach is much more difficult than the conventional approach, since the processes of technical change are very complex and empirical understanding of the determinants of technological change is still lacking.

Schumpeter (1942) distinguishes three stages in the process of technological change. Invention of a new product or process; innovation, which is the transformation of an invention into a commercial product, accomplished through continual improvement and refinement of the new product or process; diffusion, which is the process of gradually adoption of the innovation by other firms or individuals from a small niche community to being in widespread use. The technological change process is usually initiated by a public or private investment activity called research and development (R&D) in the subsequent phases of invention and innovation.<sup>1</sup> The output of the R&D activities is an intangible asset to the firms, "knowledge capital", which is used together with other inputs to generate revenues. The magnitude and direction of corporate investment in the knowledge sector is governed by private profit incentives from (at least partly) appropriable innovations. However, it is difficult to exclude others from knowledge embodied in an industry's innovation. It contributes to the innovation process of other industries and results in knowledge spillovers, or positive externalities, to competing firms (Griliches 1979). The diffusion of new, economically superior technologies is never instantaneous, but typically follows an S-shaped (sigmoid) curve that measures the rate of diffusion of innovations over time (Rogers 1995). The fraction of potential users that adapt the new technology rises only slowly in the early stage, then gets faster, then slows down again as the technology reaches maturity and approaches saturation. Experience with a technology leads to a gradual improvement over time as a function of learning processes: learning in R&D stages, learning at the manufacturing stage ("learning-by-doing") and learning as a result of use of the product ("learning by using") (Rosenberg 1982).

Virtually all studies of induced technological change in environmental economics have been theoretical (e.g. Bovenberg and Smulders 1995). Since the effects of environmental policy on the rate and direction of technological change and, in particular, the costs of reducing GHG emissions cannot be resolved at a purely theoretical level, there is a need for empirical analyses. The aim of this paper is to provide an overview of how exogenous and endogenous technological change is represented in applied environment-economy models and what implications the different specifications have for modeling results. Section 2 gives a taxonomy of different model types (bottom-up, top-down, integrated assessment models). Section 3 explores exogenous specifications of technical change such as the autonomous energy efficiency parameter, the specification of backstop technologies and technology snapshots. Section 4 discusses different approaches to incorporate induced technical progress.

For an overview of different innovation models, such as the technology-push approach or the demand-pull model, see Rothwell (1992).

Even though the theory of induced technological change is still in development at present, three main elements in models of technological innovation can be identified: (i) corporate investment in research and development (R&D) in response to market conditions, (ii) spillovers from R&D and (iii) technology learning, especially learning-by-doing (LBD). Section 5 shows how technical change is accounted for in well-known economic models of climate change. Section 6 summarizes the quantitative results obtained from modeling exercises and the implications of different specifications of technological change. Section 7 suggests different model extensions and concludes.

#### 2. Modeling approaches: Bottom-up vs. top-down

Models of complex socio-economic systems require simplifying assumptions on system boundaries and system relationships. They operate under the assumption that the underlying economic structure will remain unchanged, or else it will change in a specified way. Technological change can be integrated to economic models of climate change as it relates to this assumption. There are two broad approaches for modeling the interaction between energy, the environment and the economy. They differ mainly with respect to the emphasis placed on a detailed, technologically based treatment of the energy system, and a theoretically consistent description of the general economy. The models placing emphasis on the former are purely partial models of the energy sector, lacking interaction with the rest of the economy. In general, they are bottom-up engineering-based linear activity models with a large number of energy technologies to capture substitution of energy carriers on the primary and final energy level, process substitution, process improvements (gross efficiency improvement, emission reduction), or energy savings. They are mostly used to compute the least-cost method of meeting a given demand for final energy or energy services subject to various system constraints such as exogenous emission reduction targets. Bottom-up models embed new technologies and model the penetration of these technologies based on costs and performance characteristics. Technological change occurs as one technology is substituted by another.

The models emphasizing the above-mentioned description of the general economy are general economic models with only rudimentary treatment of the energy system. Using the top-down approach, they describe the energy system (similar to the other sectors) in a highly aggregated way by means of neoclassical production functions that capture substitution possibilities through substitution elasticities. They usually do not rely on direct descriptions of technologies. Technological change, rather, alters the costs of production at a commodity or industry level. Top-down models may be classified as open (demand driven Keynesian) or closed (general equilibrium) models (for a model classification see e.g. Weyant 1999). Macroeconometric models are based on long-run time series data. They consist of econometrically estimated equations without equilibrium assumptions. Macroeconometric models offer a lot of economic detail, but little structural detail. Because of their structural features, they are especially suited for short-run or medium-run evaluation and forecasting. Computable general equilibrium (CGE) models have become the standard tool for the analysis of the economy-wide impact of greenhouse gas abatement policies on resource allocation and the associated implications for incomes of economic agents (Grubb et al. 1993). They are well-known Arrow-Debreu models that concern the interaction of consumers and producers in markets. The household preferences as well as the production of commodities are usually captured by nested constant elasticity of substitution (CES) functions. General equilibrium provides a consistent framework for studying price-dependent interactions between the energy system and the rest of the economy. This is important, since carbon abatement policies not only cause direct adjustments on fossil fuel markets; they also produce indirect spillovers to other markets which in turn feed back to the economy (Böhringer and Löschel 2002). There are also various mixes of these broad types. Sometimes a top-down representation of the economy is linked with a bottom-up description of technologies in energy markets (Manne 1981). A synthesis of bottom-up and top-down modeling with a direct technology description in the energy sectors and top-down regular functional forms in the other production sectors is presented by Böhringer (1998).

Models for the integrated assessment of climate protection strategies, i.e. Integrated Assessment Models (IAMs), currently try to approach climate change modeling in a very comprehensive way, by gathering knowledge from diverse scientific fields. Economic models are combined with environmental or climate change sub-models. IAMs are divided into two broad categories, which vary according to the policy options available to the regulator. Policy evaluation (or simulation) IAMs evaluate the effect of an exogenous policy on biosphere, climate and economic systems. In contrast, policy optimization IAMs have the purpose of finding the efficient or cost-efficient climate change policy and simulating the effect of an efficient level of carbon abatement. Since this is a complex process, such models typically have relatively simple economic and climate sectors (IPCC 2001).

Since top-down models are based on behavioral relations they are well suited for the analysis of long-term innovation. In addition, they provide a consistent framework for the assessment of knowledge accumulation and technology spillovers between different sectors. In contrast to bottom-up models, rebound effects (Binswanger 2001) linked to the introduction of a new technology - reduced energy prices stimulate consumption - are fully integrated. Technological change in top-down models is described through the relationship of inputs and outputs. Existing technologies are gradually replaced as relative prices of alternative technologies change. Changes in technologies are the result of price substitution along a given production isoquant (described by price elasticities) and shifts of the isoquant through changes in factor demand. In contrast, technological change in bottom-up models often occurs through the sudden penetration of new technologies (snapshot technologies). New technologies are adapted rapidly, or even instantaneously, in optimizing models, because they have higher efficiencies. The rate of overall technical change depends on the difference between snapshots and the pace of adoption (Edmonds et al. 2000). Absolute shifts in bottomup models neglect transaction costs, inertia in the energy system and market failures on the demand side (e.g. information costs, high discount rates) and thus yield too optimistic cost estimates. The same is true for the inclusion of backstop technologies in top-down models (see below). Even though the specific class of a model is important for GHG mitigation cost projections, recent model studies show that the different modeling approaches are less important than model differences in assumptions about cost and baseline definition, producer and consumer flexibility and the depiction of technological change dynamics (Hourcade and Robinson 1996, Weyant 2000). Indeed, the difference in the descriptions of technological progress seems to be the most important explanation for the inequality between top-down and bottom-up models in the assessment of economic costs of GHG emissions (Carraro and Galeotti 1997).

#### 3. Exogenous specification of technological change

Basically all models of climate change agree with Solow (1956) in assuming an exogenous overall productivity growth of about 2-3 percent per year (Aznar and Dowlatabadi 1999). In addition, exogenous technological change can be introduced to any economic model of climate change by an autonomous energy efficiency (AEEI) parameter or by assumptions about future costs of energy technologies (backstop). Modelers have to make assumptions about parameter values in the former case, while they make direct assumptions about the technology in the latter.

#### 3.1. Autonomous energy efficiency improvement

The decoupling of economic growth and energy use is commonly represented by an exogenous parameter referred to as the autonomous energy efficiency improvements (see e.g. Manne and Richels 1992, Nordhaus 1994). The AEEI is a heuristic measure of all non-price driven improvements in technology, which in turn affect the energy intensity. It may be constant or follow some estimated non-linear time trends. The use of constant AEEI terms in energy-economy modeling follows the assumption that the innovation of new energy technologies is related to a large number of minor improvements which come mostly from applying results from the common, gradually evolving pool of knowledge (Jacobsen 2001).

The AEEI parameter may either represent structural changes in the economy, i.e. changes in the share of energy in total economic output over time, or sector specific technological change, i.e. changes in the energy use per unit output of an industry through time. It is simply included as a separate coefficient in the production or cost functions of the models (factor augmenting or price diminishing technical change). A constant-elasticity-of-substitution (CES) unit cost function (*c*) with exogenous price diminishing technical change for input *i* ( $\gamma_i$ ,  $\gamma_i > 0$ ) is given by:

$$c = \sum_{i} \left[ \delta_{i} \left( p_{i} \cdot e^{-\gamma_{i} \cdot t} \right)^{1-\sigma} \right]^{1/1-\sigma}, \qquad (1)$$

where  $\delta$  is the distribution parameter,  $\sigma$  is the substitution elasticity, *t* indexes time, and  $p_i$  is the input price. Due to the price effects of the energy efficiency improvement, the AEEI parameter reduces the energy-output ratio in CES production functions only if the substitution elasticity is less than unity (Kemfert and Welsch 2000).

Of course, changes in energy (carbon) intensity are also determined by the responsiveness of energy demand to changes in energy prices, i.e. the price elasticity of the demand for energy, and simple price-induced factor substitution. As the relative price of energy increases, e. g. in response to climate change policies to reduce GHG emissions, consumers and firms substitute inputs away from energy and towards others (e. g. increased use of labor instead of machines, or public transport instead of privately owned vehicles, see Maybey et al. 1997). The degree to which capital or labor can be substituted for energy is determined by the elasticity of substitution between energy and these factors, creating thus another important parameter for the development of energy intensity in global models. The carbon intensity is additionally determined by the substitution possibilities among fossil fuels (e. g. switching from coal to gas for electricity generation). However, the changes in demand for production inputs with relative prices represent choices among policy options already available. In contrast, the AEEI improves the technology available to the producers (and the consumers) and alters the production function itself. The main problem when including technical progress with the AEEI is the difficulty to distinguish between technical progress and long-term price effects (Jones 1994).

#### 3.2. Backstop technologies

Another approach to include technical progress used in many macroeconomic models is the incorporation of exogenously provided discrete new technologies. These backstop technologies are energy sources that are already known, but not yet commercial. Since the price mechanism determines the production technology used, backstop technologies come into play as, on the one hand, they mature and costs fall with technological progress and, on the other hand, production costs of conventional technologies rise, either due to the depletion of conventional energy resources or environmental policies and associated prices increases. It is commonly assumed that a backstop technology is available in unlimited supply at a constant - and usually relatively high - marginal cost, i.e. price of the backstop. The high costs of the backstop technologies reflect the associated costs of R&D investments. The upper limit to which energy prices increase is then given by the production cost of the speculative future technologies. Backstop technologies eliminate the effect of increasing energy costs reported by other models. In long-term model projections, backstop technologies often provide a large percentage of global energy by the middle of the century. Consequently, the assumed availability and cost of backstop technologies have important impacts on model outcomes (Manne and Richels 1994). Energy-economic models typically include either or both fossil and non-fossil backstop technologies. Backstop technologies are sometimes purely generic, synthetic technologies. Sometimes they are more fully specified or existing technologies such as photovoltaic or fuel cells, ethanol from biomass, nuclear fusion, and advanced fossil fuel generation technologies such as shale oil (see, for instance, Babiker et al. 1999, Manne and Richels 1992 and 1999 and Peck and Teisberg 1992 and 1999).

Backstop technologies in macroeconomic models are a special form of so-called "technology snapshots". Technology snapshot models describe in considerable detail future available technologies. Particularly bottom-up engineering models specify many alternative technologies for energy production. This can be also done in hybrid top-down and bottom up models (Böhringer 1998). The different technology snapshots substitute each other according to some economic criteria over time (Edmonds et al. 2000). Discrete technology choice models make assumptions about the technological diffusion, i.e. assumptions about the degree of penetration for existing technologies together with assumptions on individual technological progress. Exogenous technological change, specified in one of these ways, is able to assess the effect of technical progress, but it can not model aspects like innovation or diffusion. Especially, it can not consider technologies potentially developed in the future.

#### 4. Endogenous specification of technological change

#### 4.1. Investment in R&D

A newer class of modeling approaches treats innovation as a product of explicit investment in research and development. This approach is inspired by macro models of induced technological change (Romer 1990, Lucas 1988 and Grossman and Helpman 1994). The new growth theory builds on the recognition that technological innovation is an economic activity. It arises from the efforts of profit-maximizing agents within the economy and is an endogenous response to Schumpeterian profit incentives. Knowledge is explicitly treated as non-rival and not (fully) appropriable. Investment in R&D generates spillovers, or positive externalities, which allow an economy to grow infinitely. These implications of the new growth theory for technological change and long-run economic growth stand in contrast to conventional neoclassical growth models with exogenous technical change and decreasing returns to investment in physical capital in which income per capita does not grow in the steady-state. In the new growth theory literature, endogenous technological change focuses on neutral technological change and aggregate R&D expenditure, while induced technological change focuses on the direction of R&D efforts and biases in technological change (Jaffe et al. 2000). The induced innovation hypothesis is presented in climate change models by Carraro (1998), Nordhaus (1999), Goulder and Schneider (1999), Buonanno et al. (2000) and Goulder and Mattai (2000).

Goulder and Mathai (2000) use a partial equilibrium model of knowledge accumulation in which a firm chooses the time paths of abatement and R&D investment that minimize the costs of achieving a certain emission target. Induced technological progress is incorporated in the abatement cost function (C) that depends on the level of abatement (A) and on the stock of knowledge (H) (cost-function model). Knowledge accumulation and thus technological progress may either be R&D based or based on learning-by-doing (LBD) in carbon abatement. In the first case, the evolution of the knowledge stock is described by:

$$H_t = \alpha_t H_t + k \Psi (I_t, H_t), \qquad (2)$$

where *I* is investment in knowledge, i.e. R&D expenditure,  $\alpha$  is the rate of autonomous technological progress and  $\Psi$  is the knowledge accumulation function. The evolution of the knowledge stock in the latter case is given by:

$$\dot{H}_{t} = \alpha_{t}H_{t} + k\Psi(A_{t}, H_{t}).$$
(3)

The parameter k describes whether R&D-based or LBD-based induced technological progress is present. While knowledge accumulation is costly in the R&D-based case, it is free in the LBD-based representation.

Nordhaus (1999) and Buonanno et al. (2000) describe the endogenous reaction of innovations in different sectors to price variations (or regulation) in a neoclassical optimal growth framework (the RICE model, see below). The economic agent chooses the optimal level of investment and R&D effort. R&D is the source of technical change. Resources directed to R&D improve the state of knowledge (called "innovation"). The innovation-possibility frontier, i.e. the production function for new knowledge, is given by a constant-elasticity function of the level of research in the sector. This function is calibrated to empirical studies of the inventive process. It is assumed that current innovations build on past innovations ("building on shoulders" model of innovation). There are diminishing returns to inventive activity in the particular period but the innovation possibilities are replenished in the next period.

Nordhaus (1999) adds an energy/carbon input to the DICE Cobb-Douglas production function. The rate of energy efficiency improvement varies with the amount of additional R&D expenditure in the energy sector. Carbon abatement in this framework is either due to induced technological change or factor substitution. Buonanno et al. (2000) describe economic output (Q) as a function of the stock of knowledge capital ( $K_R$ ), physical capital (K) and labor (L):

$$Q = A \cdot K_R^\beta \cdot \left( L^\gamma \cdot K^{1-\gamma} \right), \tag{4}$$

where *A* indicates exogenous technological change. Since the stock of knowledge is a factor of production, changes in knowledge capital through R&D efforts raise the productivity of resources and result in non-environmental technical progress. As long as the output elasticity

of knowledge  $\beta$  is positive, the production will be characterized by increasing returns to scale, i.e. by endogenous technological change as referred to by the new growth theory. In addition, the knowledge stock affects the emission-output ratio (*E*/*Q*):

$$\frac{E}{Q} = \left[\sigma + \chi \cdot e^{-\alpha \cdot K_R}\right] \cdot \left(1 - \mu\right),\tag{5}$$

where  $\sigma$  is an exogenous parameter describing the technology in abatement,  $\mu$  is the rate of abatement effort and  $\alpha$  describes the elasticity of the emissions-output ration with respect to knowledge capital. As long as the scaling coefficient  $\chi$  is positive, R&D efforts will also result in induced environmental technological change in form of an improvement in energy-efficiency, i.e. a reduction in the influence of energy and carbon inputs on output. Thus, R&D and increased knowledge extend the productivity of the firm and reduce the negative impact on the environment.

Another economy-wide analytic approach to model induced technical change in greenhouse gas abatement is given by Goulder and Schneider (1999). They construct a dynamic general equilibrium model in which firms in each sector employ labor, physical capital and knowledge capital, fossil-based (conventional) and non-fossil based (alternative) energy, energy-intensive materials and other materials to produce output. Induced technological change is achieved – similar to (4) - through a shift term in the neoclassical production functions that is influenced by the industries' R&D. Investment in new physical capital and expenditure on R&D activities expand the respective capital stocks. Knowledge accumulation reduces the input requirements for the industries. The accumulation of knowledge is costly. The model considers the connections between the demand and the supply of R&D and, accordingly, of the scarce knowledge-generating resources (e.g. labor), the rate of technological change and the policy initiative. Goulder and Schneider emphasize the importance of accounting for opportunity costs of inducing technological change, i.e. the costs of redirecting R&D resources from one sector to another. An increase in conventional fuel prices in response to a carbon tax increases the markets for carbon free technologies and creates an incentive for increased R&D in these sectors. This induce technical change thus lowering the costs of carbon free technologies. But increased investment in R&D by one sector reduces investment in R&D by other sectors since it demands scarce knowledgegenerating resources (crowding-out). Rapid technological change in one sector will thus be accompanied by less rapid technological progress in other industries.

#### 4.2. Spillover effects

Spillover effects from investment in R&D, or positive technological externalities, provide the source for long-term growth in the macro-level new growth theory. Empirical studies demonstrate the significance of spillovers (Griliches 1992). With external economies, the social rate of return exceeds the private rate of return on investment in R&D. There is evidence that research has a social return of 30 to 70 percent per annum as compared to private returns on capital of 6 to 15 percent per annum (Nordhaus 1999). The fundamental role of spillovers makes their incorporation in the modeling of induced technological change imperative. However, in firm-level innovation theory, spillovers affect investment incentives prior to innovation, since innovations are not fully appropriable with spillovers. In this framework, spillover effects are no longer strictly positive externalities. Since the underlying mechanisms of innovation and spillovers are currently incompletely understood, Weyant and Olavson (1999) suggest the use of spillovers only as a heuristic modeling tool that accounts for the macro-level observation. Appropriability and spillovers should be separated: Firms invest in R&D given some expected approbiability of knowledge as described in the previous section. Spillovers are overlaid on this models as an add-on, strictly positive feature resulting from previous investment decisions. The level of investment in R&D is then governed by private investment incentives, but the rate of innovation with spillovers may be higher than the respective rate without. Spillovers may be intrasectoral or intersectoral, local or international.

Goulder and Schneider (1999) introduce knowledge spillovers to an individual firm in an industry by a scale factor ( $\gamma$ ) in a CES production technology that is an increasing function of (non-excludable) spillover knowledge enjoyed by all firms (*H*):

$$X = \gamma (H) \cdot \left( N^{\rho} \cdot G^{\rho} \right)^{1/\rho}, \qquad (6)$$

where X is output, N is appropriable knowledge, G is an aggregate of all other production inputs and  $\rho$  is the substitution parameter. The spillover benefit is given by

$$H_{t+1} = H_t + \beta R_t, \tag{7}$$

where *R* is the industry-wide expenditure on R&D and  $\beta$  shows the magnitude of potential spillovers. It is assumed that firms regard *H* as exogenous.

Buonanno et al. (2000) account for international knowledge spillovers by introducing the stock of world knowledge ( $K_w = \sum K_R$ ) both in the production function and in the emissions-output ratio of the DICE model:

$$Q = A \cdot K_R^\beta \cdot K_W^\varepsilon \cdot \left( L^\gamma \cdot K^{1-\gamma} \right), \tag{4'}$$

$$\frac{E}{Q} = \left[\sigma + \chi \cdot e^{-\alpha \cdot K_R - \theta \cdot K_W}\right] \cdot \left(1 - \mu\right).$$
(5')

This spillover modeling takes into account that technologies and institutional structures diffuse internationally as the world economies are increasingly linked through international trade, capital flows, and technology transfers. With spillovers, each region learns from the knowledge (technology) of the rest of the world. Advanced technologies developed in industrialized countries are gradually adopted by developing countries. Multi-regional top-down models that account explicitly for international trade flows are – in contrast to bottom-up models - generally well suited to assess these effects of geographical diffusion of new technologies (Botteon et al. 1994).

#### 4.3. Technology learning

Initial installations of technological innovations are often expensive. Costs decline as individuals, enterprises and industries gain experience with them. The learning or experience curve describes technological progress as a function of accumulating experience with the production (learning-by-doing for manufacturers) and the use (learning-by-using for consumers) of a technology during its diffusion. Technological learning has been observed historically for many different industries and is a well-established empirical concept. Several authors suggest learning curves as a meaningful presentation of technological change in global energy models (e.g. Grübler et al. 1999 and Azar and Dowlatabadi 1999). Learning-by-

doing in technologies or systems as a source of technical change was first emphasized by Arrow (1962). The Boston Consultancy Group (BCG 1968) established the experience curve concept, relating total costs and cumulative quantity. It takes into account all parameters that influence the total costs of a product such as production improvements (process innovations, learning effects and scaling efforts), product development (product innovation, product redesign, and product standardization), and decreases in process input costs (Neij 1997, IEA 2000) and traces them through technological and product evaluation. Kline and Rosenberg (1986) show that technology learning may have a larger impact on technological progress than the initial process of development itself.

In the basic model with a learning mechanism, technological progress is expressed in terms of decreasing specific costs of a technology (C) as a function of cumulative (installed) capacity (K). The cumulative capacity is used as a measure of the knowledge accumulation that occurs during the manufacturing (learning-by-doing) and the use of the technology (learning-by-using) (Christiansson 1995). A commonly used learning-by-doing function is:

$$C = \alpha \cdot K^{-\beta} \,, \tag{8}$$

where  $\alpha$  is a normalization parameter and  $\beta$  is the learning elasticity (or learning index) (Anderson 1999). With this definition, every doubling of total installed capacity reduces specific costs by a factor of  $2^{-\beta}$ . This factor is also called progress rate (PR) and defines the speed of learning. The complementary learning rate (LR = 1 - PR) gives the percentage reduction in the specific capital cost of newly installed capacity for every doubling of cumulative capacity. A recent overview of experience accumulation and cost reduction for a number of energy technologies is given by McDonald and Schrattenholzer (2001). Learning rates for different energy technologies given by IEA (2000) are summarized in Table 1. The variation is significant; there must be a 19-fold increase in the cumulative installed capacity for biomass, but only a 3-fold increase for photovoltaics in Europe for a 50 percent reduction in costs. Accordingly, the time span until the new technologies become competitive differs considerably.

Technology learning often shows different phases (Mackay and Probert 1998). In the research, development and deployment phase (RD&D), high learning rate can usually be observed as technologies first seek a market niche. In the commercialization or diffusion

phase, learning rates are lower. When market saturation is reached, the learning rate may fall close to zero and technology learning is hardly noticeable (IEA 2000). To avoid both overestimation and underestimation of the future progress ratios, the different phases of technologies must be taken into account when extrapolating historical data. Photovoltaics (mostly module) have had a learning rate of roughly 20 percent since 1975. This is also true for wind and gas turbines in their research, development and deployment phase. However, gas turbines show a learning rate of about 10 percent in the commercialization phase. The learning rates of technologies, such as photovoltaics, may also fall to 10 percent when the technology and market mature (UNDP 2000).

Energy technology	Europe	USA	Elsewhere
Photovoltaics	35	18	_
Wind power	18	32	-
Biomass (electricity)	15	-	-
Ethanol production	-	-	20
Supercritical coal	3	-	-
Natural gas-fired combined cycle	4	-	-

TT 11 1	т •	C / 1	1 • /•	()
Table 1	Learning rates	for energy techno	100100 (11	nercent)
	Loaining rails		102103 (11	

Source: IEA (2000) and own calculations.

#### 4.4. Other approaches

There have been a number of other ad hoc attempts to model technological change. Since technological progress cannot be observed directly, various econometric studies infer technological change by observing the dynamics of other variables (latent variable approach). The estimation equations describe the economic mechanism through which economic variables affect technical progress. Technical progress in this approaches is non-autonomous, but not explicitly modeled as the outcome of optimizing behavior of economic agents. Rather, technical change is often modeled as an empirical function replacing the deterministic AEEI as a proxy of technological change. The advantage is that all types and sources of technological development are included in these models (Weyant and Olavson 1999). However, since the empirical models build on historical data, they are suitable only for short-or medium term cost assessments.

Dowlatabadi and Oravetz (1996) and Dowlatabadi (1998) construct a model of priceinduced energy efficiency. The empirical analysis of the historical rate of technical change embodied in the AEEI parameter showed that it has been strongly influenced by energy price changes over the period from 1954 through 1994. The assumption of a deterministic technical progress may thus not be correct for long periods of time. They replace the exogenous AEEI by an energy efficiency response model of induced technological change that reacts to energy prices. In the period of declining energy prices up to 1974 the price induced energy efficiency improvement (PIEEI) was about –1.6 percent, but only about 1 percent in the period of rising energy prices thereafter. This confirms the intuition that price increases spur greater innovation and a greater diffusion of energy-saving technologies. Empirically, the linkage of efficiency improvement to energy prices demands a distinction between price-induced shifts in factor inputs and price-induced improvements of efficiency, which is difficult to establish.

Jorgenson and Wilcoxen (1990) present the most complete empirical model of endogenous technological change. They include technological progress by allowing input prices to interact with a time trend. For each industry, they use a translog unit cost function  $(c_i)$  in which costs depend on the prices of all inputs  $(p_j)$  and an autonomous technological (time) trend (t):

$$\ln c_i(p,t) = \alpha_o^i + \sum_j \alpha_j^i \cdot \ln p_j^t + \alpha_T^i \cdot t + \frac{1}{2} \sum_j \sum_k \beta_{jk}^i \cdot \ln p_j^t \cdot \ln p_k^t + \sum_j \gamma_{jT}^i \cdot t \cdot \ln p_j^t + \frac{1}{2} \delta_{TT}^i \cdot t^2.$$
(9)

The input coefficients are derived by Shephard's lemma. They vary to implicitly capture the effects of induced technological change:

$$\frac{x_{ji}}{x_i} = \left(\alpha_j^i + \sum_k \beta_{jk}^i \cdot \ln p_k^t + \gamma_{jT}^i \cdot t\right) \cdot \frac{c_i}{p_j}.$$
(10)

Differentiating the cost function with respect to *t* shows the components of technical progress:

$$\frac{\partial \ln c}{\partial t} = \alpha_T^i + \sum_j \gamma_{jT}^i \cdot \ln p_j^t + \delta_{TT}^i \cdot t \,. \tag{11}$$

The parameters  $\alpha_T$  and  $\delta_{TT}$  are exogenous and represent the neutral component of technical progress. Technical progress becomes endogenous because of the second term. An increase in the price of input *i* reduces cost reduction due to productivity growth if the bias of technical change is input *i* using ( $\gamma_T > 0$ ). Hence, price induced productivity growth partially endogenizes technological development ("factor price bias"). Jorgenson and Wilcoxen statistically estimated productivity growth for 35 industries from observed changes of input prices and the index of technology using extensive time series data on inter-industry transactions from 1947 to 1985. Different industries exhibit different rates of technical change. They found energy saving technical change ( $\gamma_T < 0$ ) only for three industries. The approach is often called semi-endogenous, since it does not really endogenize technical change. The biased productivity growth parameter  $\gamma_T$  is still exogenous and there is no explicit explanation for the assumed relationship.

Another approach to incorporate technical change in environmental models is the use of macroeconometric vintage models with capital vintages involving different technologies (Solow 1959). These technologies may have effects on the production function, the input structure (energy efficiency), or the flexibility of the vintage. For example, substitution possibilities among inputs of production may be higher with new vintages than with old vintages (Bergman 1990, Burniaux et al. 1992), or the parameterization of each vintage's cost function may differ (Conrad and Henseler-Unger 1986). Hence, with different vintages, the input structure not only depends on the substitution possibilities for the vintages when relative prices change but also on the input structure of the new vintage and the rate at which new investment goods disperse into the economy. If new capital brings better technology, then productivity should rise as capital investment accelerates and as the average age of the capital stock declines. Disembodied technical progress in the vintages and by adding the input coefficient of the new vintage. The price dependent input coefficient for energy based on the new relative prices in the following period is

$$\frac{E_{t+1}}{X_{t+1}} = \frac{1}{1+g} \left[ \left(1-\delta\right) \cdot \alpha_t^{\sigma} \cdot \left(\frac{PE_{t+1}}{PX_{t+1}}\right)^{-\sigma} + \left(g+\delta\right) \cdot \beta_t^{\sigma} \cdot \left(\frac{PE_{t+1}^{NVK}}{PX_{t+1}}\right)^{-\sigma} \right],\tag{12}$$

where  $\delta$  is the rate of capital depreciation, g is the growth rate of output, X is the output with price PX, E is the energy input, PE and  $PE^{NVK}$  are the energy prices associated with the respective vintage, and  $\alpha$  and  $\beta$  are distribution parameters for energy of the old and new production process, respectively (Conrad 2001). Currently, there are many established vintage models, either macroeconometric models, dynamic general equilibrium models or technically based bottom-up models. However, the vintage approach that embodies technological progress in each year's capital vintage is - except for the pace and the size of capital investment - again an exogenous specification of technological progress.

Lee et al. (1990) incorporate induced technological change following the neo-Keynesian approach by using Kaldor's technical progress function that links productivity growth and capital accumulation (Kaldor 1957). Technological progress ( $T_t$ ) simply follows a recursive relationship with gross investment and R&D expenditure:

$$T_t = \alpha_o + \alpha_1 \cdot d_t, \tag{13}$$

$$d_t = \beta d_{t-1} + (1 - \beta) \cdot \ln\left(I_t + \gamma R D_t\right). \tag{14}$$

In these equations,  $\alpha_0$  and  $\alpha_1$  are constants,  $\beta$  measures the impact of past quality-adjusted investment on the current state of technical advance (i.e. R&D and gross investment form a technological index designed to represent the stock of knowledge accumulated over time),  $\gamma$  is a weighting parameter for R&D expenditure, *I* is gross investment, and *RD* is R&D expenditure. However, this very simplistic modeling of endogenous technological change has no microeconomic underpinning (Grubb and Köhler 2000).

Carraro and Galeotti (1996 and 1997) infer technological progress from the dynamics of the capital stock (*K*), which is split into an energy-saving one ( $K_e$ ) and an energy consuming one ( $K_p$ ). A new vintage of the capital stock is added to the two components each year. The growth rate of the total capital stock (*g*) is given by:

$$g = g_p + \left[g_e - g_p\right] \cdot \frac{K_e}{K},\tag{15}$$

where  $g_e$  and  $g_p$  are the growth rates of the different capital stocks that are econometrically estimated. The amount of R&D carried out by firms is endogenously determined by relative prices, market demand, and policy variables such as environmental taxes or R&D subsidies. R&D activities affect the firms' decision to install energy-saving capital and, thereby, the composition of their capital stock. An increase in R&D expenditure is likely to produce investment in environment-friendly capital. The ratio of the stock of the environmental friendly capital to polluting capital is used as an indicator of technological progress. That is, there is a quantified relationship between R&D and technological progress. It should be noted, however, that most of the observed increase in energy efficiency is not related to the choice between energy extensive and energy intensive capital.

### 5. Economic models of environmental policy

This section surveys recent developments in technological modeling within economy wide models. As already pointed out, the incorporation of endogenous technical change is difficult and thus commonly neglected. It requires the modeling of economic processes with externalities and increasing returns to scale. Introducing endogenous technical change makes the models highly non-linear systems and significantly increases the computational complexity. The non-convex optimization raises the difficult computational problem of finding a unique optimum.

Most top-down models rely on exogenous technological change. They usually incorporate an AEEI parameter and allow for backstop technologies. Typical CGE models are GEM-E3, GREEN, PACE and MIT-EPPA. The General Equilibrium Model for Energy–Economics–Environment (GEM-E3) is a computable general equilibrium model for the European Union member states that links the macro-economy with details of the interaction with the environment and the energy system (Capros et al. 1997). Technological change in GEM-E3 is characterized by a constant AEEI parameter. The same is true for PACE, a large-scale modeling system designed to analyze the economic implications of environmental policies (Böhringer 1999), and the OECD GREEN model, which assumes an AEEI parameter

of 1 percent (Burniaux et al. 1992). In addition, the GREEN model has a so-called putty/semiputty dynamic structure. Two kinds of capital goods coexist in each period: capital installed in previous periods, and new capital as a result of investment in the current period. To include endogenous technological change, a vintage re-calibration is applied at the beginning of each period and parameters of the production structure are modified to reflect the changing composition of capital (Beghin et al. 1995). Substitution between energy and labor is more feasible in the most recent vintages. Hence, increased capital adjustment or replacement rates increase technological progress since energy can be better substituted by other inputs (Burniaux et al. 1992). The RICE integrated assessment model for climate change policy analysis (Nordhaus 1994) assumes an exponential slowdown in productivity growth rate for each region  $g_i$  over time (t):

$$g_{i}(t) = g_{i}(0) \cdot e^{-d_{j} \cdot t},$$
 (16)

where *d* is its constant rate of decline calibrated to historical growth rates of output per capita. Carbon-saving technological change is introduced with the assumption that decarbonization (decline in the global carbon-emissions-to-output ratio) is about 1 percent until 2005 and reduces to zero over subsequent decades (Nordhaus and Boyer 1999). The MIT-EPPA model considers a reference AEEI parameter of 0.75 percent and backstop technologies for oil, gas and electricity. The oil and gas technologies are hydrocarbon-based, representing coal gasification and shale oil. The electricity technology is a carbon-free renewable alternative representing a combination of solar, wind, and other technologies. There is a mark-up above the base-year cost of the substituted commercial fuel for the backstop technologies. For example, the markup for synfuel oil in the USA is 2.8, i.e. ceteris paribus the shale backstop technology becomes a competitive energy supply technology after a 2.8-fold increase in oil prices.

In the macroeconometric energy-environment-economy model for Europe (E3ME), induced technological change is modeled using Kaldor's technical progress function. E3ME is an econometric input-output model with 32 sectors and 14 EU regions (Barker and Köhler 1998). The main endogenous variables in E3ME, e. g. technological change, are determined from functions estimated on historical data for European energy use and the economy. Carraro and Galeotti (1996 and 1997) include induced technical change in the WARM model

using again a latent variable approach with respect to the capital stock. WARM is an econometric general equilibrium model estimated for 12 EU countries. In WARM, an increase in a firm's R&D expenditure, which depends inter alia on policy variables such as environmental taxation and innovation subsidies, leads to more environment-friendly capital. The factor price bias of Jorgenson and Wilcoxen (1990) is included in DGEM.

Bottom-up models are almost exclusively technology snapshot models that examine a suite of technological alternatives over time. A number of bottom-up models have integrated endogenous technological change that assumes LBD. Examples are MESSAGE (Messner and Schrattenholzer 1998, Seebregts et al. 1999) and market allocation, called MARKAL (Barreto and Kypreos 1999, Kypreos and Barreto 1998). MESSAGE and MARKAL are dynamic linear programming models of the energy sector (bottom-up), that are generally used in a tandem with the MACRO (top-down) macro-economic model which provides economic data for the energy system (Manne 1981). They optimize a choice between different technologies using given abatement costs and carbon emission targets. Both models account for the substantial uncertainty associated with the time of arrival and performance of new technologies by employing a stochastic rather than a deterministic optimization technique. Cost and performance information of a variety of advanced renewable and fossil-fuel-based alternative energy technologies are put into the model (technology snapshots). However, there is no modeling of technological change, just adoption of different available technologies. MARKAL considers 13, MESSAGE 77 technologies. Both include six learning technologies. Learning technologies in MESSAGE and MARKAL are displayed in Table 2. POLES adds other features of the innovation process. The share of technical potential that is realized as economic potential for a technology is described by a logistic curve and depends on the payback period. The payback period also influences the coefficient describing the speed of diffusion in a logistic diffusion curve (Kouvaritakis et al. 2000).

6	7
11	15
4	7
13	-
11	15
19	18
-	15
	4 13 11

#### Table 2Learning rates in MARKAL and MESSAGE (in percent)

Source: Seebregts et al. (1999).

Dowlatabadi and Oravetz (1997) implemented the PIEEI relationship in the Integrated Climate Assessment Model (ICAM 3). ICAM 3 is a non-optimizing, sequential, stochastic decision model with a fairly complex structure. Endogenous technological change ( $T_t$ ) in ICAM is written as:

$$T_{t} = A(y) + E\left[\alpha \cdot \frac{p_{t} - p_{t-1}}{p_{t-1}}\right],$$
(17)

where A is the base energy efficiency as a function of per capita income y,  $\alpha$  is a constant coefficient and  $p_t$  is the average price of energy. Expectations of energy price increases lead to technological innovation and diffusion. The autonomous energy efficiency parameter A accounts for the historical fact that energy intensity has dropped in most sectors even under falling energy prices. Dowlatabadi (1998) added LBD and included endogenous technological progress not only for energy efficiency in conversion and end-uses, but also in the discovery and recovery of oil and gas. The integrated model to assess the greenhouse effect (IMAGE) also uses AEEI and PIEEI to endogenize technological change. In IMAGE, the AEEI parameter is assumed to decline exponentially and is linked to the capital turnover rate (Alcamo et al. 1998). Nordhaus (1999) incorporates induced innovation as a function of the

R&D-based knowledge stock and spillover effects in the R&DICE model. The model then determines carbon taxes, capital investment and the level of R&D in the energy/carbon sector that maximizes discounted per-capita consumption. FUND, another IAM, deals with technological development through running different scenarios about technological progress (Tol 1999). Table 3 summarizes the treatment of technical change in well-known climate change models.

Model	Type <sup>*</sup>	Representation of TC**	Reference
DGEM	CGE	Factor price bias	Jorgenson and Wilcoxen (1990)
DICE/RICE	IAM	AEEI	Nordhaus (1994)
E3ME	ME	Latent variable (Investment)	Barker and Köhler (1998)
ETC-RICE	CGE	R&D, Spillovers	Buonanno et al. (2000)
FUND	IAM	TS, Scenarios	Tol (1999)
GEM-E3	CGE	AEEI	Capros et al. (1997)
GOULDER	ME	LBD, R&D, Spillovers	Goulder and Mathai (2000)
	CGE	R&D, Spillovers	Goulder and Schneider (1999)
GREEN	CGE	AEEI, Vintages	Burniaux et al. (1992)
ICAM3	IAM	LBD, PIEEI, TS	Dowlabadi (1998)
IMAGE	IAM	AEEI, PIEEI	Alcamo et al. (1998)
MARKAL	ES	LBD, TS	Barreto and Kypreos (1999)
MESSAGE	ES	LBD, TS	Grübler and Messner (1998)
MIT-EPPA	CGE	AEEI, Backstops	Babiker et al. (2001)
PACE	CGE	AEEI, Backstops	Böhringer (1999)
POLES	ES	LBD, Diffusion curves	Kouvaritakis et al. (2000)
R&DICE	IAM	R&D, Spillovers	Nordhaus (1999)
WARM	ME	Latent variable (Capital stock)	Carraro and Galeotti (1997)

 Table 3
 Technical change characteristics in energy-economy models

\* CGE: computational general equilibrium model, ES: energy system model,

IAM: integrated assessment model, ME: macroeconometric model

<sup>\*\*</sup> AEEI: autonomous energy efficiency improvements, LBD: learning-by-doing, PIEEI: price induced energy efficiency improvements, TS: technology snapshot,

#### 6. Implications of exogenous and induced technical change

Long-term projections are very sensitive to differences in assumed AEEI rates. Even small changes in the AEEI parameter result in large differences for energy demand and emissions in the baseline and, hence, the total costs of emissions reductions (Manne and Richels 1990, Böhringer et al. 2000). The higher (lower) the AEEI is, the lower (higher) the baseline emissions and lower (higher) the costs to reach a climate target relative to a given base year are. There may be uncertainty as to the true value for the AEEI parameter. Given the multiple factors that influence changes in energy consumption over time, it is difficult to capture the appropriate value. Estimates for AEEI rates vary widely, ranging from 0.4 per cent to 1.5 per cent (Grubb et al. 1993 and Weyant 1999). Assumptions about (backstop) technologies have large impacts on mitigation costs. Hourcade and Robinson (1996) showed that emissions reduction costs decrease over time, simply because more technologies become available. The magnitude of this effect depends on the characteristics of the assumed backstop technologies. The importance of technology assumptions for the costs of achieving different atmospheric concentrations is backed by Edmonds et al. (1997).

The omission of induced technological change may lead to an overestimation of abatement costs and the trend increase in emissions. Price induced technical change as studied by Dowlatabadi and Oravetz (1997) spurs technical progress as energy prices increase. This leads to a decline in energy use and carbon emissions per unit of energy thereby lowering emissions and substantially reducing the associated costs of mitigation. Dowlatabadi (1998) added LBD, which further reduces abatement costs. Carbon taxes may lead to a change to low carbon technologies from the given menu of alternative technologies, which in turn may speed up the diffusion and bring down the costs of the new energy technologies (Azar and Dowlatabadi 1999). However, if technological progress reduces the cost of energy consumption, endogenous technical change increases business as usual emissions. Therefore, additional abatement efforts are required to meet some specified target, but the associated costs will also be lower (Dowlatabadi 1998).

Quantitative results from bottom-up models with relatively cheap low- or non-carbon technologies and LBD indicate low gross costs of abatement. Long-run stabilization of emissions may – without accounting for transaction costs - even be achieved at no cost (Anderson et al. 2000). The structure of energy production and the optimal technology mix

26

implemented changes considerably with LBD. While standard coal and nuclear power are still the dominant technologies in 2050 in the static case, advanced coal, new nuclear and solar technologies dominate the mix in the LBD case. Investment costs are reduced by about 20 percent in the LBD case in 2050 (Messner and Schrattenholzer 1998). Goulder and Mathai (2000) find that costs and the required (optimal) carbon tax are generally lower with the existence of induced technological progress than without, although the impacts of induced technical change are rather weak. The modest effects of learning-by-doing carbon abatement in aggregated models contrast the results from bottom-up approaches. This result may be traced back to the lack of sectoral disaggregation in these model approaches, which do not take into account the varying impacts of LBD on different industries at different stages of development. Conventional energy industries tend to be mature industries, in which learningby-doing effects may be rather small in contrast e. g. to renewable energy industries (Anderson 1999).

Unlike learning-by-doing, investment in R&D is associated with opportunity costs. Models that take into account the opportunity costs of investment in R&D like those by Goulder and Schneider (1999), Nordhaus (1999) and Buonanno et al. (2000) find only weak impacts of induced technological change on the gross costs of abatement. Goulder and Schneider (1999) claim that the opportunity cost of redirecting limited R&D resources to the energy sector steepens the decline in GDP associated with the introduction of a carbon tax under induced technological change, i.e. a given tax leads to larger gross costs (25 percent higher GDP loss). The reason for this is that in equilibrium, the rate of return on R&D is equalized across sectors and equals the rate of return to other investments. An increase in R&D expenses on renewables with induced technological change leads to reduced R&D and, hence, reduced productivity in other sectors. Technical expertise (human capital) is "crowded out" from other applications. The gross costs rise for a given carbon tax. Since induced technological change, on the other hand, results in lower costs for the alternative technologies, greater abatement is achieved and environmental benefits increase. Induced technical change can thus significantly raise the benefits of a given policy, but can not reduce the costs substantially. It is shown that the additional benefits outweigh the higher gross costs, yielding higher net benefits with induced technological change for a given tax than without technological change. If a target is set, the required tax and, hence, gross costs are lower. Subsidies on industry-specific R&D lead to over-investment in R&D. However, R&D market

failures (knowledge spillovers), which lead to under-investment in R&D by firms, justify R&D subsidies as a second policy instrument besides the carbon tax. Large spillovers to the benefit of non-energy related industries strengthen the argument for subsidization to overcome the R&D market failure. Carraro and Galeotti (1996 and 1997) also promote a policy mix of subsidies for environmentally friendly R&D along with taxes. A counter-example, building on the crowding out argument, is given by Kverndokk et al. (2000).

In the same vein, Nordhaus (1999) finds that the opportunity cost of R&D (and LBD) is a severe limitation to the effects of induced technological change. The effects of spillovers, which provide free progress (as does LBD) and thus lessen the negative impacts of environmental policies when considered in the models (Goulder and Mathai 2000), are also reduced by opportunity cost considerations, since high spillovers of R&D lead to high opportunity costs of redirecting R&D to the energy sector. The costliness of R&D makes the influence of induced technical change considerations insignificant in the R&DICE model. Energy R&D increases by less than 2 percent per decade, which reduces the emission-output ratio by 0.0075 percent per decade and the carbon intensity of energy use in 2100 by about 0.5 percent relative to the base path. Induced innovation seems to be a much less powerful factor in implementing climate-change policies than substitution of labor and capital for energy, which accounts for approximately 99 percent of changes in emissions, concentrations, and temperature.

Grübler and Messner (1998) point out that models with an AEEI typically result in a deferral of investment decisions until the technology has become cheap enough to be competitive. Even though absolute abatement requirement increases with late action, it is still better to wait as abatement becomes cheaper. With induced technological change the derivation of the shape of the least-cost mitigation pathway becomes more complex (Grubb 1997). Goulder and Mathai (2000) find that with R&D-induced technological change, some abatement is shifted from the present to the future. Schneider and Goulder (1997) state the case for the early introduction of carbon taxes, but not necessarily abatement. Wigley et al. (1996) argue for postponed action. In bottom-up models with LBD, early emissions-reduction measures are more preferable since they generate knowledge that lowers the relative costs of future abatement (Anderson 1999, Grübler and Messner 1998, Grubb et al. 1995). The effect of LBD-induced technological change on the time profile of abatement costs in the cost-function model of Goulder and Mathai (2000) is ambiguous.

In the policy debate of climate change, spillovers from Annex-B countries' abatement to non-abating developing countries play an important role. Models with exogenous technological change find negative environmental spillovers from abatement. Unilateral abatement policies will result in negative spillovers from abatement as abatement action by one country is partly offset by increased emissions in other countries when energy intensive industries move (leakage effect) (Böhringer and Löschel 2002). In a more dynamic perspective, induced technological change influences the leakage effects through the international diffusion of endogenous technological progress, i.e. technology spillovers as introduced by Buonanno et al. (2000). In the presence of induced technological change, cleaner technologies developed as a response to abatement policies in industrialized countries may internationally diffuse generating positive spillovers for non-abating countries. This effect arises in ICAM (Dowlatabadi and Oravetz 1997). The diffusion of cleaner technologies from emission constrained regions outweighs the classical components of leakage and also results in reduced emission elsewhere. The reason for this is the enormous leverage effect exerted by diffusion of new technologies over decades (Grubb 2000).

The implications of the different approaches to represent technological change in economic climate change models for the costs of environmental policy, the appropriate policy instruments, the timing of mitigation action and international spillovers (leakage) are summarized in Table 4.

Implications	Exogenous TC	Endogenous TC
Costs of environmental policy	High / low (depending on assumptions)	Low / high (opportunity costs)
Policy instrument	Tax	Tax / subsidy
Timing implications	Defer abatement	Accelerate abatement
Spillover / leakage	Negative spillover / positive leakage	Positive spillover / negative leakage

 Table 4
 Effects of technological change in climate-energy models

#### 7. Conclusion

This survey presents different approaches to represent technological change and the limitations inherent in existing economy-environment models. It highlights the importance of understanding the process of technological change in global climate change modeling, since the direction and extent of technological change proves crucial for the environmental impact of future economic activities. Quantifying the impacts of price-induced technological change, learning-by-doing and learning-by-using, and exogenous technological change poses a major challenge. The current state-of-the-art modeling of (endogenous) technological change still relies heavily on ad-hoc assumptions. Future extensions of the presented approaches should be targeted towards a further endogenization and an improved realism in the modeling of the process of technological change. Most important seems to be the incorporation of path-dependence and inertia, the uncertainty in major innovations, the discontinuity in the process of technological change, and the heterogeneity in firm behavior and investment incentives (Weyant and Olavson 1999). In addition, the need for calibration of R&D related model parameters requests extended econometric studies to provide an empirical background for modeling.

Climate policy seems highly sensitive to the dynamics of technical change, namely to path-sensitivity and lock-in phenomena (Grubb et al. 1995, Ha-Duong et al. 1997). The lockin effect is mainly determined by direct investment costs as in the case of learning, but also by other considerations, such as uncertainty or infrastructure and networks that create de facto standards for a technology (Arthur 1989). Path-dependency means that the sequence of historical events influences future possibilities. Firms may find it more profitable in the short run to invest in technologies that are already competitive, or, given high rates of technical change, investors may just wait and see. Hence, many innovations never reach commercialization even though they are potentially superior to existing, established alternatives. These behavioral and strategic issues lead to a highly path-dependent technology adoption and a path of technological change that evolves with a great deal of inertia. The models can account for path-sensitivity and lock-in effects by a careful inclusion of LBD, time lags and assumptions about the diffusion of innovations.

Path discontinuities arise through gradual progress and major innovations. The processes of technological change are characterized by a substantial uncertainty associated

30

with the time of arrival and performance of new technologies. It is, therefore, not possible to know what specific technologies will be successful in the future. This emphasizes the need of incorporating uncertainty in the modeling process, for example through the use of stochastic optimization techniques, the consideration of risk attitudes of decision makers and the incorporation of a very wide range of energy technologies and resources (Grübler et al. 1999 and Anderson 1999).

Another important aspect of the innovation process not yet accounted for in energyeconomy models is the heterogeneity in firm behavior and innovation incentives. Different firms respond differently to environmental policies. Additionally, new technologies are developed by an innovative firm and are not available to all firms, but diffuse over time. Heterogeneity is also addressed in a number of recent game theoretical studies related to strategic R&D investments (Ulph 1997). It is not easy to capture these evolutionary firm level processes in a neoclassical framework. In addition, top-down models do not provide the degree of disaggregation for firm level analysis, while bottom-up models are not well suited to study strategic considerations. Given these model limitations, current applied models look at homogenous, "representative" firms or industries and include spillovers and diffusion as demonstrated. An interesting approach to account for heterogeneity, though not heterogeneity of firms, but more general heterogeneity of industries and technologies, is the hybrid modeling that synthesize bottom-up and top-down approaches in energy-economic models.

#### References

- Alcamo, J., E. Kreileman and R. Leemans (1998), 'Global Change Scenarios of the 21st Century, Results from the IMAGE 2.1 Model', London: Elseviers Science.
- Anderson, D. (1999), 'Technical Progress and Pollution Abatement: An Economic Review of Selected Technologies and Practices', *Imperial College Working Paper*, London.
- Anderson, D., M. Grubb and J. Köhler (2000), 'Induced Technical Change in Energy/Environmental Modeling: Analytic Approaches and Implications', *Imperial College Working Paper*, London.
- Arrow, K. (1962), 'The Economic Implications of Learning-by-Doing', *Review of Economic Studies*, 29, 155-173.
- Arthur, B. (1989), 'Competing Technologies, Increasing Returns, and Lock-in by Historical Small Events', *Economic Journal*, **99**, 116-131.
- Azar, C. and H. Dowlatabadi (1999), 'A Review of the Treatment of Technical Change in Energy Economics Models', Annual Review of Energy and the Environment, 24, 513-544.
- Babiker, M. H., J. M. Reilly, M. Mayer, R. S. Eckaus, I. S. Wing and R. C. Hyman (2001),
  'The MIT Emissions Prediction and Policy Analysis (EPPA) Model: Revisions,
  Sensitivities, and Comparisons of Results', *MIT Joint Program on the Science and Policy of Global Change Report*, **71**.
- Barker, T. and J. Köhler (1998), 'Equity and Ecotax Reform in the EU: Achieving a 10% Reduction in CO2 Emissions Using Excise Duties', *Fiscal Studies*, **19** (4), 375-402.
- Barreto, L. and S. Kypreos (1999), 'Technological Learning in Energy Models: Experience and Scenario Analysis with MARKAL and the ERIS Model Prototype', *PSI Report 99-08*.
- Beghin, J., D. Roland-Holst and D. van der Mensbrugghe (1995), 'Trade Liberalization and the Environment in the Pacific Basin: Coordinated Approaches to Mexican Trade and Environment Policy', *American Journal of Agricultural Economics*, **77**, 778-785.
- Bergman, L. (1990), 'Energy and Environmental Constraints on Growth: A CGE Modeling Approach', *Journal of Policy Modeling*, **12** (4), 671-691.
- Binswanger, M. (2001), 'Technological Progress and Sustainable Development: What About the Rebound Effect?', *Ecological Economics*, **36** (1), 119–132.

- Böhringer, C. (1998), 'The Synthesis of Bottom-Up and Top-Down in Energy Policy Modeling', *Energy Economics*, **20** (3), 234-248.
- Böhringer, C. (1999), 'PACE Policy Assessment based on Computable Equilibrium, ein flexibles Modellsystem zur gesamtwirtschaftlichen Analyse von wirtschaftspolitischen Maßnahmen', ZEW Dokumentationen, Mannheim.
- Böhringer, C., J. Jensen and T. F. Rutherford (2000), 'Energy Market Projections and Differentiated Carbon Abatement in the European Union', in C. Carraro (ed), *Efficiency* and Equity of Climate Change Policy, Dordrecht: Kluwer Academic Publishers, 199-220.
- Böhringer, C. and A. Löschel (2002), 'Economic Impacts of Carbon Abatement Strategies', in
  C. Böhringer, M. Finus and C. Vogt (eds.), *Controlling Global Warming Perspectives* from Economics, Game Theory and Public Choice, Cheltenham: Edward Elgar, (New Horizons in Environmental Economics) (forthcoming).
- Boston Consulting Group (1968), Perspectives on Experience, Boston Consulting Group Inc.
- Botteon, M., C. Carraro and M. Galeotti (1994), Endogenous Technical Change in Econometric Models of Environmental Policy: Issues and Proposed Strategies, *Nota di Lavoro* 79.94.
- Bovenberg, A. L. and S. Smulders (1995), 'Environmental Quality and Pollution-augmenting Technological Change in a Two-sector Endogenous Growth Model', *Journal of Public Economics*, 57, 369-391.
- Buonanno, P., C. Carraro and M. Galeotti (2001), 'Endogenous Induced Technical Change and the Costs of Kyoto', *FEEM NOTE di LAVORO 64.01*.
- Burniaux, J. M., J. P. Martin, G. Nicoletti and J. Oliveira (1992), 'GREEN- A Multi-Region
  Dynamic General Equilibrium Model for Quantifying the Costs of Curbing CO2
  Emissions: A Technical Manual', *Working Papers Economics and Statistics*Department 116, OECD, Paris.
- Capros, P., T. Georgakopoulos, D. Van Regemorter, S. Proost, T. F. N. Schmidt and K. Conrad (1997), 'European Union the GEM-E3 General Equilibrium Model', *Economic & Financial Modelling*, Special Double Issue, 4 (2/3).
- Carraro, C. (1998), 'Climate Modelling and Policy Strategies. The Role of Technical Change and Uncertainty', *Energy Economics*, **20**, 463-471.

- Carraro, C. and M. Galeotti (1996), 'WARM: A European Model for Energy and Environmental Analysis', *Environmental Modeling and Assessment*, **1**, 171-189.
- Carraro, C. and M. Galeotti (1997), 'Economic Growth, International Competitiveness and Environmental Protection: R & D and Innovation Strategies with the WARM Model', *Energy Economics*, **19** (1), 2-28.
- Christiansson, L. (1995), 'Diffusion and Learning Curves of Renewable Energy Technologies', *Working Paper WP-95-126*, International Institute for Applied Systems Analysis, Laxenburg, Austria.
- Conrad, K. (2001), 'Computable General Equilibrium Models in Environmental and Resource Economics', *Beiträge zur angewandten Wirtschaftsforschung 601-01*, Institut für Volkswirtschaftslehre und Statistik, Mannheim University, Mannheim.
- Conrad, K. and I. Henseler-Unger (1986), 'Applied General Equilibrium Modelling for Longterm Energy Policy in the Fed. Rep. of Germany', *Journal of Policy Modelling*, 8 (4), 531-549.
- Dowlatabadi, H. (1998), 'Sensitivity of Climate Change Mitigation Estimates to Assumptions About Technical Change', *Energy Economics*, **20** (5-6), 473-493.
- Dowlatabadi, H. and M. Oravetz (1997), 'Is There Autonomous Energy Efficiency Improvement?', *Department of Engineering and Public Policy*, Carnegie Mellon University.
- Edmonds, J., J. M. Roop and M. J. Scott (2000), 'Technology and the Economics of Climate Change Policy', *Pew Center on Global Climate Change Report*, Batelle, Washington, D. C.
- Edmonds, J., M. Wise and J. Dooley (1997), 'Atmospheric Stabilization and the Role of Energy Technology', in C. E. Walker, M. A. Bloomfield and M. Thorning (eds.), *Climate Change Policy, Risk Prioritization and U.S. Economic Growth*, American Council for Capital Formation, Washington, D.C., 71-94.
- EMF (Energy Modeling Forum) (1996), 'Markets for Energy Efficiency', *EMF Report 13*, Volume I, Stanford University.
- Glueck, H. and S. P. Schleicher (1995), Endogenous Technical Progress Induced by CO2 Reduction Policies: Simulation Results for Austria, *Environmental and Resource Economics*, 5 (2), 151-163.

- Goulder, L. H. and K. Mathai (2000), 'Optimal CO2 Abatement in the Presence of Induced Technological Change', *Journal of Environmental Economics and Management*, **39** (1), 1-38.
- Goulder, L. H. and S. Schneider (1999), 'Induced Technological Change, Crowding out, and the Attractiveness of CO2 Emissions Abatement', *Resource and Environmental Economics*, **21** (3-4), 211-253.
- Griliches, Z. (1979), 'Issues in Assessing the Contribution of Research and Development to Productivity Growth', *Bell Journal of Economics*, **10** (1), 92-116.
- Griliches, Z. (1992), 'The Search for R&D Spillovers', Scandinavian Journal of Economics, 94, 29-47.
- Grossman, G. and E. Helpman (1994), 'Endogenous Innovation in the Theory of Growth', Journal of Economic Perspectives, 8, 23-44.
- Grubb, M. (1997), 'Technologies, Energy Systems, and the Timing of CO2 Abatement: An Overview of Economic Issues', *Energy Policy*, 25, 159-172.
- Grubb, M. (2000), 'Economic Dimensions of Technological and Global Responses to the Kyoto Protocol', *Journal of Economic Studies*, **27** (1/2), 111-125.
- Grubb, M. and J. Köhler (2000), Induced Technical Change: Evidence and Implications for Energy-Environmental Modelling and Policy, *Report to the Deputy Secretary-General* Organisation for Economic Cooperation and Development, Paris.
- Grubb, M., T. Chapuis and M. Ha-Duong (1995), 'The Economics of Changing Course: Implications of Adaptability and Inertia for Optimal Climate Policy', *Energy Policy*, 23 (4/5), 417-431.
- Grubb, M., J. Edmonds, P. ten Brink and M. Morrison (1993), 'The Costs of Limiting Fossilfuel CO2 Emissions: A Survey and Analysis', Annual Review of Energy and Environment, 18, 397-478.
- Grübler, A. and S. Messner (1998), 'Technological Change and the Timing of Mitigation Measures', *Energy Economics*, **20** (5-6), 495-512.
- Grübler, A., N. Nakicenovic and D. G. Victor (1999), 'Modeling Technological Change: Implications for the Global Environment', Annual Review of Energy and the Environment, 24, 545-569.
- Ha-Duong, M., M. Grubb and J-C. Hourcade (1997), 'Influence of Socio-Economic Inertia and Uncertainty on Optimal CO2 Emission Abatement', *Nature*, **390**, 270-273.

- Hourcade, J. C. and J. Robinson (1996), 'Mitigating Factors: Assessing the Costs of Reducing GHG Emissions', *Energy Policy*, **24** (10/11), 863-873.
- IEA (International Energy Agency) (2000), *Experience Curves for Energy Technology Policy*, Paris.
- IPCC (International Panel on Climate Change) (1996), Climate Change 1995. Economic and Social Dimensions of Climate Change, Contribution of Working Group III to the Second Assessment Report of the Intergovernmental Panel on Climate Change, Cambridge: Cambridge University Press.
- IPCC (International Panel on Climate Change) (2001), Climate Change 2001: Mitigation, Contribution of Working Group III to the Third Assessment Report of the Intergovernmental Panel on Climate Change, Cambridge: Cambridge University Press.
- Jacobsen, H. K. (2001), 'Technological Progress and Long-term Energy Demand: A Survey of Recent Approaches and a Danish Case', *Energy Policy*, **29** (2), 147-157
- Jaffe, B., R. G. Newell and R. N. Stavins (2000), 'Technological Change and the Environment', *RFF Discussion Paper 00–47*, Resources for the Future, Washington, D.C.
- Jones, C. T. (1994), Accounting for Technical Progress in Aggregate Energy Demand, *Energy Economics*, **16** (4), 245-252.
- Jorgenson, D. W. and P. J. Wilcoxen (1990), 'Intertemporal General Equilibrium Modeling of U.S. Environmental Regulation', *Journal of Policy Modeling*, 12, 715-744.
- Kaldor, N. (1957), 'A Model of Economic Growth', Economic Journal, 67, 591-624.
- Kemfert, C. and H. Welsch (2000), 'Energy-Capital-Labor Substitution and the Economic Effects of CO2 Abatement: Evidence for Germany', *Journal of Policy Modeling*, 22, 641-660.
- Kemp, R. (1997), *Environmental Policy and Technical Change*: A Comparison of the Technological Impact of Policy Instruments, Edward Elgar: Cheltenham, UK.
- Kline, S. J. and N. Rosenberg (1986), 'An Overview of Innovation', in B. A. Lundvall and N.
  Rosenberg (eds.), *The Positive Sum Strategy: Harnessing Technology for Economic Growth*, Washington D.C.: National Academy Press.
- Kouvaritakis, N., A. Soria and S. Isoard (2000), 'Modelling Energy Technology Dynamics: Methodology for Adaptive Expectations Models with Learning by Doing and Learning by Searching', *International Journal of Global Energy Issues*, 14, 104–115.

- Kverndokk, S., K. E. Rosendahl and T. F. Rutherford (2000), 'Climate Policies and Induced Technological Change: Which to Choose the Carrot or the Stick?', *University of Colorado Discussion Paper*, Boulder.
- Kypreos, S. and L. Barreto (1998), A Simple Global Electricity MARKAL Model with Endogenous Learning, Paper presented at the Joint ALEP/ETSAP Workshop, Antalya, Turkey, 26-28 October 1998.
- Lee, K., M. H. Pesaran and R.G. Pierse (1990), 'Aggregation Bias in Labour Demand Equations for the UK Economy', in T. Barker and M. H. Pesaran (eds.), *Disaggregation in Econometric Modelling*, London: Routledge.
- Lucas, R. E. (1988), 'On the Mechanics of Economic Development', *Journal of Monetary Economics*, **100**, 223-251.
- Mabey, N., S. Hall, C. Smith and S. Gupta (1997), Argument in the Greenhouse: The International Economics of Controlling Global Warming, London: Routledge.
- Mackay, R. M. and S. D. Probert (1998), 'Likely Market-Penetration of Renewable-Energy Technologies', *Applied Energy*, **59** (1), 1-38.
- Manne, A. S. (1981), *ETA-MACRO: A User's Guide*, EA-1724, Electric Power Research Institute, Palo Alto, CA.
- Manne, A. S. and R. G. Richels (1990), 'The Costs of Reducing CO<sub>2</sub> Emission A Further Sensitivity Analysis', *Energy Journal*, **11** (4), 69-78.
- Manne, A. S. and R. G. Richels (1992), Buying Greenhouse Insurance: The Economic Costs of CO<sub>2</sub> Emission Limits, Cambridge, Mass: MIT Press.
- Manne, A. S. and R. G. Richels (1994), 'The Costs of Stabilizing Global CO2 Emissions: A Probabilistic Analysis Based on Expert Judgements', *Energy Journal*, **15** (1), 31-56.
- Manne, A. S. and R. G. Richels (1999), 'The Kyoto Protocol: A Cost Effective Strategy for Meeting Environmental Objectives?', *Energy Journal*, Special Issue, 1-23.
- McDonald, A. and L. Schrattenholzer (2001), Learning Rates for Energy Technologies, Energy Policy, 29 (4), 255-261.
- Messner, S. and L. Schrattenholzer (1998), Experiences From Including Technology Progress Dynamics into the Global Optimization Model MESSAGE, ECS Contribution to the Mid-Term Assessment Report of the TEEM Project, IIASA, Laxenburg, Austria.
- Neij, L. (1997), 'Use of Experience Curves to Analyze the Prospects for Diffusion and Adoption of Renewable Energy Technology', *Energy Policy*, **23** (13), 1099-1107.

- Nordhaus, W. (1994), *Managing the Global Commons: The Economics of Climate Change*, Cambridge, Mass: MIT Press.
- Nordhaus, W. (1999), *Modeling Induced Innovation in Climate-Change Policy*, Paper presented at the Workshop on Induced Technological Change and the Environment, IIASA, Laxenberg, Austria, June 1999.
- Nordhaus, W. D. and J. Boyer (1999), Roll the DICE Again: The Economics of Global Warming, Yale University.
- Peck, S. C. and T. J. Teisberg (1992), 'CETA: A Model for Carbon Emissions Trajectory Assessment', *The Energy Journal*, **13**, 55-77.
- Peck, S. C. and T. J. Teisberg (1999), 'CO2 Emissions Control Agreements: Incentives for Regional Participation', *Energy Journal*, Special Issue, 367-390.
- Rogers, E. (1995), The Diffussion of Innovation, 4th ed, New York: Free Press.
- Romer, P. M. (1990), 'Endogenous Technical Change', *Journal of Political Economy*, **98** (5), 71-102.
- Rosenberg, N. (1982), *Inside the Black Box: Technology and Economics*, Cambridge University Press.
- Rothwell, R. (1992), 'Successful Industrial Innovation: Critical Factors for the 1990s', *R&D* Management, **22** (3), 221-239.
- Schneider, S. H. and L. H. Goulder (1997), Achieving Low-Cost Emissions Targets, *Nature*, **389**, 13-14.
- Schumpeter, J. (1942), Capitalism, Socialism and Democracy, New York: Harper.
- Seebregts, A. J., T. Kram, G. J. Schaeffer, A. Stoffer, S. Kypreos, L. Barreto, S. Messner and L. Schrattenholzer (1999), 'Endogenous Technological Change in Energy System Models, Synthesis of Experience with ERIS, MARKAL and MESSAGE', *ECN-C-99-*025, ECN Netherlands Energy Research Foundation.
- Solow, R. M. (1956), 'A Contribution to the Theory of Economic Growth', *Quarterly Journal of Economics*, **70**, 71-102.
- Solow, R. (1959), 'Investment and Technological Progress', in K. Arrow, S. Karlin and P. Suppes (eds.), *Mathematical Methods in the Social Sciences*, Stanford, CA: Stanford University Press, 89-104.
- Tol, R. S. J. (1999), 'Spatial and Temporal Efficiency in Climate Policy: Applications of FUND', *Environmental and Resource Economics*, **14** (1), 33-49.

- Toman, M. (1998), Research Frontiers in the Economics of Climate Change, *Environmental* and Resource Economics, **11** (3-4), 603-621.
- Ulph, D. (1997), Environmental Policy and Technological Innovation, in: C. Carraro and D. Siniscalco (Eds.), *New directions in the economic theory of the environment*, Cambridge University Press.
- UNDP (United Nations Development Programme) (2000), World Energy Assessment: Energy and the Challenge of Sustainability, New York.
- Weyant, J. (ed) (1999), 'The Costs of the Kyoto Protocol: A Multi-Model Evaluation', *The Energy Journal*, Special Issue.
- Weyant, J. (2000), 'The Economy and Global Climate Change: An Introduction to the Economics of Climate Change Policy', *Pew Center on Global Climate Change Report*, Airlington, VA.
- Weyant, J. and T. Olavson (1999), 'Issues in Modeling Induced Technological Change in Energy, Environment, and Climate Policy', *Journal of Environmental Management and Assessment*, 1, 67-85.
- Wigley, T. M. L., R. Richels and J. A. Edmonds (1996), Economic and Environmental Choices in the Stabilisation of Atmospheric CO2 Concentrations, *Nature*, **379**, 240-243.

## NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI Fondazione Eni Enrico Mattei Working Papers Series Our working papers are available on the Internet at the following addresses: Server WWW: WWW.FEEM.IT

Anonymous FTP: FTP.FEEM.IT

To order any of these papers, please fill out the form at the end of the list.

SUST SUST	1.2001 2.2001	Inge MAYERES and Stef PROOST: <u>Should Diesel Cars in Europe be Discouraged?</u> Paola DORIA and Davide PETTENELLA: <u>The Decision Making Process in Defining and Protecting</u>		
		<u>Critical Natural Capital</u>		
CLIM	3.2001	Alberto PENCH: Green Tax Reforms in a Computable General Equilibrium Model for Italy		
CLIM	4.2001	Maurizio BUSSOLO and Dino PINELLI: Green Taxes: Environment, Employment and Growth		
CLIM	5.2001	Marco STAMPINI: Tax Reforms and Environmental Policies for Italy		
ETA	6.2001	Walid OUESLATI: Environmental Fiscal Policy in an Endogenous Growth Model with Human Capital		
CLIM	7.2001	Umberto CIORBA, Alessandro LANZA and Francesco PAULI: Kyoto Commitment and Emission Trading:		
MGMT	8.2001	<u>a European Union Perspective</u> Brian SLACK (xlv): <u>Globalisation in Maritime Transportation</u> : <u>Competition, uncertainty and</u> <u>implications for port development strategy</u>		
VOL	9.2001	Giulia PESARO: Environmental Voluntary Agreements: A New Model of Co-operation Between		
VOI	10 2001	Public and Economic Actors		
VOL	10.2001	Cathrine HAGEM: Climate Policy, Asymmetric Information and Firm Survival		
ETA	11.2001	Sergio CURRARINI and Marco MARINI: <u>A Sequential Approach to the Characteristic Function and the</u>		
	10 0001	Core in Games with Externalities		
ETA	12.2001	Gaetano BLOISE, Sergio CURRARINI and Nicholas KIKIDIS: Inflation and Welfare in an OLG Economy		
KNIOW	12 2001	with a Privately Provided Public Good		
KNOW	13.2001	Paolo SURICO: <u>Globalisation and Trade: A "New Economic Geography" Perspective</u>		
ETA	14.2001	Valentina BOSETTI and Vincenzina MESSINA: Quasi Option Value and Irreversible Choices		
CLIM	15.2001	Guy ENGELEN (xlii): Desertification and Land Degradation in Mediterranean Areas: from Science to		
CLICT	1 ( 0001	Integrated Policy Making		
SUST	16.2001	Julie Catherine SORS: Measuring Progress Towards Sustainable Development in Venice: A		
OLIOT	17 0001	Comparative Assessment of Methods and Approaches		
SUST	17.2001	Julie Catherine SORS: Public Participation in Local Agenda 21: A Review of Traditional and Innovative Tools		
	10 0001			
CLIM	18.2001	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism		
CLIM VOL	18.2001 19.2001	Johan ALBRECHT and Niko GOBBIN: <u>Schumpeter and the Rise of Modern Environmentalism</u> Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): <u>Participation Incentives and the Design</u>		
VOL	19.2001	Johan ALBRECHT and Niko GOBBIN: <u>Schumpeter and the Rise of Modern Environmentalism</u> Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): <u>Participation Incentives and the Design</u> of Voluntary Agreements		
VOL ETA	19.2001 20.2001	Johan ALBRECHT and Niko GOBBIN: <u>Schumpeter and the Rise of Modern Environmentalism</u> Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): <u>Participation Incentives and the Design</u> <u>of Voluntary Agreements</u> Paola ROTA: <u>Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs</u>		
VOL ETA ETA	19.2001 20.2001 21.2001	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design of Voluntary Agreements Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (S,s) Rule		
VOL ETA	19.2001 20.2001	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design         of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (5,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete		
VOL ETA ETA ETA	19.2001 20.2001 21.2001 22.2001	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design         of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (5,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete         Decision Processes		
VOL ETA ETA	19.2001 20.2001 21.2001	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design         of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (S,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete         Decision Processes         Stefano BOSI, Guillaume GIRMANS and Michel GUILLARD: Optimal Privatisation Design and Financial		
VOL ETA ETA ETA PRIV	19.2001 20.2001 21.2001 22.2001 23.2001	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design         of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (S,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete         Decision Processes         Stefano BOSI, Guillaume GIRMANS and Michel GUILLARD: Optimal Privatisation Design and Financial Markets		
VOL ETA ETA ETA	19.2001 20.2001 21.2001 22.2001	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design         of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (S,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete         Decision Processes         Stefano BOSI, Guillaume GIRMANS and Michel GUILLARD: Optimal Privatisation Design and Financial         Markets         Giorgio BRUNELLO, Claudio LUPI, Patrizia ORDINE, and Maria Luisa PARISI: Beyond National		
VOL ETA ETA ETA PRIV KNOW	19.2001 20.2001 21.2001 22.2001 23.2001 24.2001	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design         of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (S,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete         Decision Processes         Stefano BOSI, Guillaume GIRMANS and Michel GUILLARD: Optimal Privatisation Design and Financial         Markets         Giorgio BRUNELLO, Claudio LUPI, Patrizia ORDINE, and Maria Luisa PARISI: Beyond National         Institutions: Labour Taxes and Regional Unemployment in Italy		
VOL ETA ETA ETA PRIV	19.2001 20.2001 21.2001 22.2001 23.2001	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design         of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (S,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete         Decision Processes         Stefano BOSI, Guillaume GIRMANS and Michel GUILLARD: Optimal Privatisation Design and Financial         Markets         Giorgio BRUNELLO, Claudio LUPI, Patrizia ORDINE, and Maria Luisa PARISI: Beyond National         Institutions: Labour Taxes and Regional Unemployment in Italy         Klaus CONRAD: Locational Competition under Environmental Regulation when Input Prices and		
VOL ETA ETA PRIV KNOW ETA	<ol> <li>19.2001</li> <li>20.2001</li> <li>21.2001</li> <li>22.2001</li> <li>23.2001</li> <li>24.2001</li> <li>25.2001</li> </ol>	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design         of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (S,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete         Decision Processes         Stefano BOSI, Guillaume GIRMANS and Michel GUILLARD: Optimal Privatisation Design and Financial         Markets         Giorgio BRUNELLO, Claudio LUPI, Patrizia ORDINE, and Maria Luisa PARISI: Beyond National         Institutions: Labour Taxes and Regional Unemployment in Italy         Klaus CONRAD: Locational Competition under Environmental Regulation when Input Prices and         Productivity Differ		
VOL ETA ETA ETA PRIV KNOW	19.2001 20.2001 21.2001 22.2001 23.2001 24.2001	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design         of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (S,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete         Decision Processes         Stefano BOSI, Guillaume GIRMANS and Michel GUILLARD: Optimal Privatisation Design and Financial         Markets         Giorgio BRUNELLO, Claudio LUPI, Patrizia ORDINE, and Maria Luisa PARISI: Beyond National         Institutions: Labour Taxes and Regional Unemployment in Italy         Klaus CONRAD: Locational Competition under Environmental Regulation when Input Prices and         Productivity Differ         Bernardo BORTOLOTTI, Juliet D'SOUZA, Marcella FANTINI and William L. MEGGINSON: Sources of		
VOL ETA ETA PRIV KNOW ETA	<ol> <li>19.2001</li> <li>20.2001</li> <li>21.2001</li> <li>22.2001</li> <li>23.2001</li> <li>24.2001</li> <li>25.2001</li> </ol>	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design         of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (S,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete         Decision Processes         Stefano BOSI, Guillaume GIRMANS and Michel GUILLARD: Optimal Privatisation Design and Financial         Markets         Giorgio BRUNELLO, Claudio LUPI, Patrizia ORDINE, and Maria Luisa PARISI: Beyond National         Institutions: Labour Taxes and Regional Unemployment in Italy         Klaus CONRAD: Locational Competition under Environmental Regulation when Input Prices and         Productivity Differ         Bernardo BORTOLOTTI, Juliet D'SOUZA, Marcella FANTINI and William L. MEGGINSON: Sources of         Performance Improvement in Privatised Firms: A Clinical Study of the Global Telecommunications		
VOL ETA ETA PRIV KNOW ETA PRIV	19.2001 20.2001 21.2001 22.2001 23.2001 24.2001 25.2001 26.2001	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design         of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (S,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete         Decision Processes         Stefano BOSI, Guillaume GIRMANS and Michel GUILLARD: Optimal Privatisation Design and Financial         Markets         Giorgio BRUNELLO, Claudio LUPI, Patrizia ORDINE, and Maria Luisa PARISI: Beyond National         Institutions: Labour Taxes and Regional Unemployment in Italy         Klaus CONRAD: Locational Competition under Environmental Regulation when Input Prices and         Productivity Differ         Bernardo BORTOLOTTI, Juliet D'SOUZA, Marcella FANTINI and William L. MEGGINSON: Sources of         Performance Improvement in Privatised Firms: A Clinical Study of the Global Telecommunications         Industry		
VOL ETA ETA PRIV KNOW ETA	<ol> <li>19.2001</li> <li>20.2001</li> <li>21.2001</li> <li>22.2001</li> <li>23.2001</li> <li>24.2001</li> <li>25.2001</li> </ol>	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design         of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (S,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete         Decision Processes         Stefano BOSI, Guillaume GIRMANS and Michel GUILLARD: Optimal Privatisation Design and Financial         Markets         Giorgio BRUNELLO, Claudio LUPI, Patrizia ORDINE, and Maria Luisa PARISI: Beyond National         Institutions: Labour Taxes and Regional Unemployment in Italy         Klaus CONRAD: Locational Competition under Environmental Regulation when Input Prices and         Productivity Differ         Bernardo BORTOLOTTI, Juliet D'SOUZA, Marcella FANTINI and William L. MEGGINSON: Sources of         Performance Improvement in Privatised Firms: A Clinical Study of the Global Telecommunications		
VOL ETA ETA PRIV KNOW ETA PRIV	19.2001 20.2001 21.2001 22.2001 23.2001 24.2001 25.2001 26.2001	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design         of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (S,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete         Decision Processes         Stefano BOSI, Guillaume GIRMANS and Michel GUILLARD: Optimal Privatisation Design and Financial         Markets         Giorgio BRUNELLO, Claudio LUPI, Patrizia ORDINE, and Maria Luisa PARISI: Beyond National         Institutions: Labour Taxes and Regional Unemployment in Italy         Klaus CONRAD: Locational Competition under Environmental Regulation when Input Prices and         Productivity Differ         Bernardo BORTOLOTTI, Juliet D'SOUZA, Marcella FANTINI and William L. MEGGINSON: Sources of         Performance Improvement in Privatised Firms: A Clinical Study of the Global Telecommunications         Industry		
VOL ETA ETA PRIV KNOW ETA PRIV	19.2001 20.2001 21.2001 22.2001 23.2001 24.2001 25.2001 26.2001	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (S,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete         Decision Processes         Stefano BOSI, Guillaume GIRMANS and Michel GUILLARD: Optimal Privatisation Design and Financial Markets         Giorgio BRUNELLO, Claudio LUPI, Patrizia ORDINE, and Maria Luisa PARISI: Beyond National Institutions: Labour Taxes and Regional Unemployment in Italy         Klaus CONRAD: Locational Competition under Environmental Regulation when Input Prices and Productivity Differ         Bernardo BORTOLOTTI, Juliet D'SOUZA, Marcella FANTINI and William L. MEGGINSON: Sources of Performance Improvement in Privatised Firms: A Clinical Study of the Global Telecommunications Industry         Frédéric BROCHIER and Emiliano RAMIERI: Climate Change Impacts on the Mediterranean Coastal Zones         Nunzio CAPPUCCIO and Michele MORETTO: Comments on the Investment-Uncertainty Relationship		
VOL ETA ETA PRIV KNOW ETA PRIV CLIM	19.2001 20.2001 21.2001 22.2001 23.2001 24.2001 25.2001 26.2001 27.2001	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (S,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete         Decision Processes         Stefano BOSI, Guillaume GIRMANS and Michel GUILLARD: Optimal Privatisation Design and Financial Markets         Giorgio BRUNELLO, Claudio LUPI, Patrizia ORDINE, and Maria Luisa PARISI: Beyond National Institutions: Labour Taxes and Regional Unemployment in Italy         Klaus CONRAD: Locational Competition under Environmental Regulation when Input Prices and Productivity Differ         Bernardo BORTOLOTTI, Juliet D'SOUZA, Marcella FANTINI and William L. MEGGINSON: Sources of Performance Improvement in Privatised Firms: A Clinical Study of the Global Telecommunications Industry         Frédéric BROCHIER and Emiliano RAMIERI: Climate Change Impacts on the Mediterranean Coastal Zones         Nunzio CAPPUCCIO and Michele MORETTO: Comments on the Investment-Uncertainty Relationship in a Real Option Model		
VOL ETA ETA PRIV KNOW ETA PRIV CLIM	19.2001 20.2001 21.2001 22.2001 23.2001 24.2001 25.2001 26.2001 27.2001	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (S,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete         Decision Processes         Stefano BOSI, Guillaume GIRMANS and Michel GUILLARD: Optimal Privatisation Design and Financial Markets         Giorgio BRUNELLO, Claudio LUPI, Patrizia ORDINE, and Maria Luisa PARISI: Beyond National Institutions: Labour Taxes and Regional Unemployment in Italy         Klaus CONRAD: Locational Competition under Environmental Regulation when Input Prices and Productivity Differ         Bernardo BORTOLOTTI, Juliet D'SOUZA, Marcella FANTINI and William L. MEGGINSON: Sources of Performance Improvement in Privatised Firms: A Clinical Study of the Global Telecommunications Industry         Frédéric BROCHIER and Emiliano RAMIERI: Climate Change Impacts on the Mediterranean Coastal Zones         Nunzio CAPPUCCIO and Michele MORETTO: Comments on the Investment-Uncertainty Relationship		
VOL ETA ETA PRIV KNOW ETA PRIV CLIM ETA	19.2001 20.2001 21.2001 22.2001 23.2001 24.2001 25.2001 26.2001 27.2001 28.2001	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design         of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (S,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete         Decision Processes         Stefano BOSI, Guillaume GIRMANS and Michel GUILLARD: Optimal Privatisation Design and Financial         Markets         Giorgio BRUNELLO, Claudio LUPI, Patrizia ORDINE, and Maria Luisa PARISI: Beyond National         Institutions: Labour Taxes and Regional Unemployment in Italy         Klaus CONRAD: Locational Competition under Environmental Regulation when Input Prices and         Productivity Differ         Bernardo BORTOLOTTI, Juliet D'SOUZA, Marcella FANTINI and William L. MEGGINSON: Sources of         Performance Improvement in Privatised Firms: A Clinical Study of the Global Telecommunications         Industry         Frédéric BROCHIER and Emiliano RAMIERI: Climate Change Impacts on the Mediterranean Coastal         Zones         Nunzio CAPPUCCIO and Michele MORETTO: Comments on the Investment-Uncertainty Relationship         in a Real Option Model         Giorgio BRUNELLO: Absolute Risk Aversion and the Returns to Education		
VOL ETA ETA PRIV KNOW ETA PRIV CLIM ETA KNOW	<ol> <li>19.2001</li> <li>20.2001</li> <li>21.2001</li> <li>22.2001</li> <li>23.2001</li> <li>24.2001</li> <li>25.2001</li> <li>26.2001</li> <li>27.2001</li> <li>28.2001</li> <li>29.2001</li> </ol>	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design         of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (S,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete         Decision Processes         Stefano BOSI, Guillaume GIRMANS and Michel GUILLARD: Optimal Privatisation Design and Financial         Markets         Giorgio BRUNELLO, Claudio LUPI, Patrizia ORDINE, and Maria Luisa PARISI: Beyond National         Institutions: Labour Taxes and Regional Unemployment in Italy         Klaus CONRAD: Locational Competition under Environmental Regulation when Input Prices and         Productivity Differ         Bernardo BORTOLOTTI, Juliet D'SOUZA, Marcella FANTINI and William L. MEGGINSON: Sources of         Prédrime Ronce Improvement in Privatised Firms: A Clinical Study of the Global Telecommunications         Industry         Frédéric BROCHIER and Emiliano RAMIERI: Climate Change Impacts on the Mediterranean Coastal         Zones         Nunzio CAPPUCCIO and Michele MORETTO: Comments on the Investment-Uncertainty Relationship         in a Real Option Model         Giorgio BRUNELLO: Absolute Risk Aversion and the Returns to Education		
VOL ETA ETA PRIV KNOW ETA PRIV CLIM ETA KNOW CLIM	19.2001 20.2001 21.2001 22.2001 23.2001 24.2001 25.2001 26.2001 27.2001 28.2001 29.2001 30.2001	Johan ALBRECHT and Niko GOBBIN: Schumpeter and the Rise of Modern Environmentalism         Rinaldo BRAU, Carlo CARRARO and Giulio GOLFETTO (xliii): Participation Incentives and the Design         of Voluntary Agreements         Paola ROTA: Dynamic Labour Demand with Lumpy and Kinked Adjustment Costs         Paola ROTA: Empirical Representation of Firms' Employment Decisions by an (S,s) Rule         Paola ROTA: What Do We Gain by Being Discrete? An Introduction to the Econometrics of Discrete         Decision Processes         Stefano BOSI, Guillaume GIRMANS and Michel GUILLARD: Optimal Privatisation Design and Financial         Markets         Giorgio BRUNELLO, Claudio LUPI, Patrizia ORDINE, and Maria Luisa PARISI: Beyond National         Institutions: Labour Taxes and Regional Unemployment in Italy         Klaus CONRAD: Locational Competition under Environmental Regulation when Input Prices and         Productivity Differ         Bernardo BORTOLOTTI, Juliet D'SOUZA, Marcella FANTINI and William L. MEGGINSON: Sources of         Performance Improvement in Privatised Firms: A Clinical Study of the Global Telecommunications         Industry         Frédéric BROCHIER and Emiliano RAMIERI: Climate Change Impacts on the Mediterranean Coastal         Zones         Nunzio CAPPUCCIO and Michele MORETTO: Comments on the Investment-Uncertainty Relationship         in a Real Option Model         Giorgio BRUNELLO: Absolute Risk Aversion and the Returns to Education		

Coalition		
Theory	33.2001	Shlomo WEBER and Hans WIESMETH: From Autarky to Free Trade: The Impact on Environment
Network		
ETA	34.2001	Margarita GENIUS and Elisabetta STRAZZERA: Model Selection and Tests for Non Nested Contingent
NRM	35.2001	Valuation Models: An Assessment of Methods Carlo GIUPPONI: The Substitution of Hazardous Molecules in Production Processes: The Atrazine
	00.2001	Case Study in Italian Agriculture
KNOW	36.2001	Raffaele PACI and Francesco PIGLIARU: Technological Diffusion, Spatial Spillovers and Regional
	27 2001	<u>Convergence in Europe</u>
PRIV CLIM	37.2001 38.2001	Bernardo BORTOLOTTI: Privatisation, Large Shareholders, and Sequential Auctions of Shares Barbara BUCHNER: What Really Happened in The Hague? Report on the COP6, Part I, 13-25
CEIIII	00.2001	November 2000, The Hague, The Netherlands
PRIV	39.2001	Giacomo CALZOLARI and Carlo SCARPA: Regulation at Home, Competition Abroad: A Theoretical
KNOW	40.0001	Framework
KNOW Coalition	40.2001 41.2001	<i>Giorgio BRUNELLO</i> : <u>On the Complementarity between Education and Training in Europe</u> <i>Alain DESDOIGTS and Fabien MOIZEAU</i> (xlvi): <u>Multiple Politico-Economic Regimes, Inequality and</u>
Theory	41.2001	Growth
Network		
Coalition	42.2001	Parkash CHANDER and Henry TULKENS (xlvi): Limits to Climate Change
Theory Network		
Coalition	43.2001	Michael FINUS and Bianca RUNDSHAGEN (xlvi): Endogenous Coalition Formation in Global Pollution
Theory		Control
Network		
Coalition Theory	44.2001	<i>Wietze LISE, Richard S.J. TOL and Bob van der ZWAAN</i> (xlvi): <u>Negotiating Climate Change as a Social</u> <u>Situation</u>
Network		
NRM	45.2001	Mohamad R. KHAWLIE (xlvii): The Impacts of Climate Change on Water Resources of Lebanon-
	46 2001	Eastern Mediterranean
NRM	46.2001	<i>Mutasem EL-FADEL and E. BOU-ZEID</i> (xlvii): <u>Climate Change and Water Resources in the Middle</u> <u>East: Vulnerability, Socio-Economic Impacts and Adaptation</u>
NRM	47.2001	Eva IGLESIAS, Alberto GARRIDO and Almudena GOMEZ (xlvii): An Economic Drought Management
		Index to Evaluate Water Institutions' Performance Under Uncertainty and Climate Change
CLIM	48.2001	Wietze LISE and Richard S.J. TOL (xlvii): Impact of Climate on Tourist Demand
CLIM	49.2001	Francesco BOSELLO, Barbara BUCHNER, Carlo CARRARO and Davide RAGGI: <u>Can Equity Enhance</u> <u>Efficiency? Lessons from the Kyoto Protocol</u>
SUST	50.2001	Roberto ROSON (xlviii): Carbon Leakage in a Small Open Economy with Capital Mobility
SUST	51.2001	Edwin WOERDMAN (xlviii): Developing a European Carbon Trading Market: Will Permit Allocation
CLICT	FO 2001	Distort Competition and Lead to State Aid?
SUST SUST	52.2001 53.2001	<i>Richard N. COOPER</i> (xlviii): <u>The Kyoto Protocol: A Flawed Concept</u> <i>Kari KANGAS</i> (xlviii): <u>Trade Liberalisation, Changing Forest Management and Roundwood Trade in</u>
0001	00.2001	Europe
SUST	54.2001	Xueqin ZHU and Ekko VAN IERLAND (xlviii): Effects of the Enlargement of EU on Trade and the
CLICT	FF 2001	Environment
SUST	55.2001	M. Ozgur KAYALICA and Sajal LAHIRI (xlviii): <u>Strategic Environmental Policies in the Presence of</u> Foreign Direct Investment
SUST	56.2001	Savas ALPAY (xlviii): <u>Can Environmental Regulations be Compatible with Higher International</u>
		Competitiveness? Some New Theoretical Insights
SUST	57.2001	Roldan MURADIAN, Martin O'CONNOR, Joan MARTINEZ-ALER (xlviii): Embodied Pollution in
SUST	58.2001	<u>Trade:</u> Estimating the "Environmental Load Displacement" of Industrialised Countries Matthew R. AUER and Rafael REUVENY (xlviii): Foreign Aid and Direct Investment: Key Players in the
5651	50.2001	Environmental Restoration of Central and Eastern Europe
SUST	59.2001	Onno J. KUIK and Frans H. OOSTERHUIS (xlviii): Lessons from the Southern Enlargement of the EU
	(0.0001	for the Environmental Dimensions of Eastern Enlargement, in particular for Poland
ETA	60.2001	<i>Carlo CARRARO, Alessandra POME and Domenico SINISCALCO</i> (xlix): <u>Science vs. Profit in Research:</u> Lessons from the Human Genome Project
CLIM	61.2001	Efrem CASTELNUOVO, Michele MORETTO and Sergio VERGALLI: Global Warming, Uncertainty and
		Endogenous Technical Change: Implications for Kyoto
PRIV	62.2001	Gian Luigi ALBANO, Fabrizio GERMANO and Stefano LOVO: On Some Collusive and Signaling
CLIM	63.2001	<u>Equilibria in Ascending Auctions for Multiple Objects</u> Elbert DIJKGRAAF and Herman R.J. VOLLEBERGH: <u>A Note on Testing for Environmental Kuznets</u>
C.1101	00.2001	Curves with Panel Data

CLIM	64.2001	Paolo BUONANNO, Carlo CARRARO and Marzio GALEOTTI: Endogenous Induced Technical Change
		and the Costs of Kyoto
CLIM	65.2001	Guido CAZZAVILLAN and Ignazio MUSU (I): Transitional Dynamics and Uniqueness of the Balanced-
	(( 0001	Growth Path in a Simple Model of Endogenous Growth with an Environmental Asset
CLIM	66.2001	<i>Giovanni BAIOCCHI and Salvatore DI FALCO</i> (I): <u>Investigating the Shape of the EKC: A Nonparametric</u>
CLIM	67.2001	<u>Approach</u> Marzio GALEOTTI, Alessandro LANZA and Francesco PAULI (1): <u>Desperately Seeking (Environmental)</u>
CLIN	67.2001	Kuznets: A New Look at the Evidence
CLIM	68.2001	Alexey VIKHLYAEV (xlviii): The Use of Trade Measures for Environmental Purposes – Globally and in
CLIW	00.2001	the EU Context
NRM	69.2001	Gary D. LIBECAP and Zeynep K. HANSEN (li): U.S. Land Policy, Property Rights, and the Dust Bowl of
		the 1930s
NRM	70.2001	Lee J. ALSTON, Gary D. LIBECAP and Bernardo MUELLER (li): Land Reform Policies, The Sources of
		Violent Conflict and Implications for Deforestation in the Brazilian Amazon
CLIM	71.2001	Claudia KEMFERT: Economy-Energy-Climate Interaction – The Model WIAGEM -
SUST	72.2001	Paulo A.L.D. NUNES and Yohanes E. RIYANTO: Policy Instruments for Creating Markets for
		Bodiversity: Certification and Ecolabeling
SUST	73.2001	Paulo A.L.D. NUNES and Erik SCHOKKAERT (lii): Warm Glow and Embedding in Contingent
		Valuation
SUST	74.2001	Paulo A.L.D. NUNES, Jeroen C.J.M. van den BERGH and Peter NIJKAMP (lii): Ecological-Economic
		Analysis and Valuation of Biodiversity
VOL	75.2001	Johan EYCKMANS and Henry TULKENS (li): Simulating Coalitionally Stable Burden Sharing
		Agreements for the Climate Change Problem
PRIV	76.2001	Axel GAUTIER and Florian HEIDER: What Do Internal Capital Markets Do? Redistribution vs.
	<b>FF 2</b> 001	Incentives
PRIV	77.2001	Bernardo BORTOLOTTI, Marcella FANTINI and Domenico SINISCALCO: Privatisation around the
ET A	79 2001	World: New Evidence from Panel Data
ETA	78.2001	<i>Toke S. AIDT and Jayasri DUTTA</i> (li): <u>Transitional Politics. Emerging Incentive-based Instruments in</u> Environmental Regulation
ETA	79.2001	Alberto PETRUCCI: Consumption Taxation and Endogenous Growth in a Model with New
LIA	79.2001	Generations
ETA	80.2001	<i>Pierre LASSERRE and Antoine SOUBEYRAN</i> (li): <u>A Ricardian Model of the Tragedy of the Commons</u>
ETA	81.2001	Pierre COURTOIS, Jean Christophe PÉREAU and Tarik TAZDAÏT: An Evolutionary Approach to the
		Climate Change Negotiation Game
NRM	82.2001	Christophe BONTEMPS, Stéphane COUTURE and Pascal FAVARD: Is the Irrigation Water Demand
		Really Convex?
NRM	83.2001	Unai PASCUAL and Edward BARBIER: A Model of Optimal Labour and Soil Use with Shifting
		Cultivation
CLIM	84.2001	Jesper JENSEN and Martin Hvidt THELLE: What are the Gains from a Multi-Gas Strategy?
CLIM	85.2001	Maurizio MICHELINI (liii): IPCC "Summary for Policymakers" in TAR. Do its results give a scientific
		support always adequate to the urgencies of Kyoto negotiations?
CLIM	86.2001	Claudia KEMFERT (liii): Economic Impact Assessment of Alternative Climate Policy Strategies
CLIM	87.2001	Cesare DOSI and Michele MORETTO: Global Warming and Financial Umbrellas
ETA	88.2001	Elena BONTEMPI, Alessandra DEL BOCA, Alessandra FRANZOSI, Marzio GALEOTTI and Paola ROTA:
		Capital Heterogeneity: Does it Matter? Fundamental Q and Investment on a Panel of Italian Firms
ETA	89.2001	Efrem CASTELNUOVO and Paolo SURICO: Model Uncertainty, Optimal Monetary Policy and the
CL II (	00.0001	Preferences of the Fed
CLIM	90.2001	Umberto CIORBA, Alessandro LANZA and Francesco PAULI: Kyoto Protocol and Emission Trading:
	01 2001	Does the US Make a Difference?
CLIM	91.2001	Zhong Xiang ZHANG and Lucas ASSUNCAO: <u>Domestic Climate Policies and the WTO</u> Anna ALBERINI, Alan KRUPNICK, Maureen CROPPER, Nathalie SIMON and Joseph COOK (lii): <u>The</u>
SUST	92.2001	Willingness to Pay for Mortality Risk Reductions: A Comparison of the United States and Canada
SUST	93.2001	Riccardo SCARPA, Guy D. GARROD and Kenneth G. WILLIS (lii): Valuing Local Public Goods with
5051	<i>J</i> 5.2001	Advanced Stated Preference Models: Traffic Calming Schemes in Northern England
CLIM	94.2001	Ming CHEN and Larry KARP: Environmental Indices for the Chinese Grain Sector
CLIM	95.2001	Larry KARP and Jiangfeng ZHANG: Controlling a Stock Pollutant with Endogenous Investment and
		Asymmetric Information
ETA	96.2001	Michele MORETTO and Gianpaolo ROSSINI: On the Opportunity Cost of Nontradable Stock Options
SUST	97.2001	Elisabetta STRAZZERA, Margarita GENIUS, Riccardo SCARPA and George HUTCHINSON: The Effect of
		Protest Votes on the Estimates of Willingness to Pay for Use Values of Recreational Sites
NRM	98.2001	Frédéric BROCHIER, Carlo GIUPPONI and Alberto LONGO: Integrated Coastal Zone Management in
		the Venice Area - Perspectives of Development for the Rural Island of Sant'Erasmo

NRM	99.2001	Frédéric BROCHIER, Carlo GIUPPONI and Julie SORS: Integrated Coastal Management in the Venice		
		Area - Potentials of the Integrated Participatory Management Approach		
NRM	100.2001			
		A Methodological Framework		
PRIV	101.2001	Enrico C. PEROTTI and Luc LAEVEN: Confidence Building in Emerging Stock Markets		
CLIM	102.2001	Barbara BUCHNER, Carlo CARRARO and Igor CERSOSIMO: On the Consequences of the U.S.		
		Withdrawal from the Kyoto/Bonn Protocol		
SUST	UST 103.2001 Riccardo SCARPA, Adam DRUCKER, Simon ANDERSON, Nancy FERRAES-EHUAN, V			
		Carlos R. RISOPATRON and Olga RUBIO-LEONEL: Valuing Animal Genetic Resources in Peasant		
		Economies: The Case of the Box Keken Creole Pig in Yucatan		
SUST	104.2001	R. SCARPA, P. KRISTJANSON, A. DRUCKER, M. RADENY, E.S.K. RUTO, and J.E.O. REGE: Valuing		
		Indigenous Cattle Breeds in Kenya: An Empirical Comparison of Stated and Revealed Preference		
		Value Estimates		
SUST	105.2001	Clemens B.A. WOLLNY: The Need to Conserve Farm Animal Genetic Resources Through Community-		
		Based Management in Africa: Should Policy Makers be Concerned?		
SUST	106.2001	J.T. KARUGIA, O.A. MWAI, R. KAITHO, Adam G. DRUCKER, C.B.A. WOLLNY and J.E.O. REGE:		
		Economic Analysis of Crossbreeding Programmes in Sub-Saharan Africa: A Conceptual Framework		
		and Kenyan Case Study		
SUST	107.2001	W. AYALEW, J.M. KING, E. BRUNS and B. RISCHKOWSKY: Economic Evaluation of Smallholder		
		Subsistence Livestock Production: Lessons from an Ethiopian Goat Development Program		
SUST	108.2001	Gianni CICIA, Elisabetta D'ERCOLE and Davide MARINO: Valuing Farm Animal Genetic Resources by		
		Means of Contingent Valuation and a Bio-Economic Model: The Case of the Pentro Horse		
SUST	109.2001	Clem TISDELL: Socioeconomic Causes of Loss of Animal Genetic Diversity: Analysis and Assessment		
SUST	110.2001	M.A. JABBAR and M.L. DIEDHOU: Does Breed Matter to Cattle Farmers and Buyers? Evidence from		
		West Africa		
SUST	1.2002	K. TANO, M.D. FAMINOW, M. KAMUANGA and B. SWALLOW: Using Conjoint Analysis to Estimate		
		Farmers' Preferences for Cattle Traits in West Africa		
ETA	2.2002	Efrem CASTELNUOVO and Paolo SURICO: What Does Monetary Policy Reveal about Central Bank's		
		Preferences?		
WAT	3.2002	Duncan KNOWLER and Edward BARBIER: The Economics of a "Mixed Blessing" Effect: A Case Study		
<i></i>		<u>of the Black Sea</u>		
CLIM	4.2002	Andreas LÖSCHEL: Technological Change in Economic Models of Environmental Policy: A Survey		

(xlii) This paper was presented at the International Workshop on "Climate Change and Mediterranean Coastal Systems: Regional Scenarios and Vulnerability Assessment" organised by the Fondazione Eni Enrico Mattei in co-operation with the Istituto Veneto di Scienze, Lettere ed Arti, Venice, December 9-10, 1999.

(xliii)This paper was presented at the International Workshop on "Voluntary Approaches, Competition and Competitiveness" organised by the Fondazione Eni Enrico Mattei within the research activities of the CAVA Network, Milan, May 25-26,2000.

(xliv) This paper was presented at the International Workshop on "Green National Accounting in Europe: Comparison of Methods and Experiences" organised by the Fondazione Eni Enrico Mattei within the Concerted Action of Environmental Valuation in Europe (EVE), Milan, March 4-7, 2000

(xlv) This paper was presented at the International Workshop on "New Ports and Urban and Regional Development. The Dynamics of Sustainability" organised by the Fondazione Eni Enrico Mattei, Venice, May 5-6, 2000.

(xlvi) This paper was presented at the Sixth Meeting of the Coalition Theory Network organised by the Fondazione Eni Enrico Mattei and the CORE, Université Catholique de Louvain, Louvain-la-Neuve, Belgium, January 26-27, 2001

(xlvii) This paper was presented at the RICAMARE Workshop "Socioeconomic Assessments of Climate Change in the Mediterranean: Impact, Adaptation and Mitigation Co-benefits", organised by the Fondazione Eni Enrico Mattei, Milan, February 9-10, 2001 (xlviii) This paper was presented at the International Workshop "Trade and the Environment in the Perspective of the EU Enlargement", organised by the Fondazione Eni Enrico Mattei, Milan, May 17-18, 2001

(xlix) This paper was presented at the International Conference "Knowledge as an Economic Good", organised by Fondazione Eni Enrico Mattei and The Beijer International Institute of Environmental Economics, Palermo, April 20-21, 2001

(l) This paper was presented at the Workshop "Growth, Environmental Policies and + Sustainability" organised by the Fondazione Eni Enrico Mattei, Venice, June 1, 2001 (li) This paper was presented at the Fourth Toulouse Conference on Environment and Resource Economics on "Property Rights, Institutions and Management of Environmental and Natural Resources", organised by Fondazione Eni Enrico Mattei, IDEI and INRA and sponsored by MATE, Toulouse, May 3-4, 2001

(lii) This paper was presented at the International Conference on "Economic Valuation of Environmental Goods", organised by Fondazione Eni Enrico Mattei in cooperation with CORILA, Venice, May 11, 2001

(liii) This paper was circulated at the International Conference on "Climate Policy – Do We Need a New Approach?", jointly organised by Fondazione Eni Enrico Mattei, Stanford University and Venice International University, Isola di San Servolo, Venice, September 6-8, 2001

# 2001 SERIES

MGMT	Corporate Sustainable Management (Editor: Andrea Marsanich)
CLIM	Climate Change Modelling and Policy (Editor: Marzio Galeotti)
PRIV	Privatisation, Antitrust, Regulation (Editor: Bernardo Bortolotti)
KNOW	Knowledge, Technology, Human Capital (Editor: Dino Pinelli)
NRM	Natural Resources Management (Editor: Carlo Giupponi)
SUST	Sustainability Indicators and Environmental Evaluation (Editor: Marialuisa Tamborra)
VOL	Voluntary and International Agreements (Editor: Carlo Carraro)
ETA	Economic Theory and Applications (Editor: Carlo Carraro)

## 2002 SERIES

MGMT	Corporate Sustainable Management (Editor: Andrea Marsanich)
CLIM	Climate Change Modelling and Policy (Editor: Marzio Galeotti)
PRIV	Privatisation, Antitrust, Regulation (Editor: Bernardo Bortolotti)
KNOW	Knowledge, Technology, Human Capital (Editor: Dino Pinelli)
NRM	Natural Resources Management (Editor: Carlo Giupponi)
SUST	Sustainability Indicators and Environmental Evaluation (Editor: Marialuisa Tamborra)
VOL	Voluntary and International Agreements (Editor: Carlo Carraro)
ETA	Economic Theory and Applications (Editor: Carlo Carraro)

#### SUBSCRIPTION TO "NOTE DI LAVORO"

Starting from January 1998 Fondazione Eni Enrico Mattei issues a Periodic E-mail "Note di Lavoro" Bulletin listing the titles and the abstracts of its most recent Working Papers.

All the "Note di Lavoro" listed in the Bulletin are available on the Internet and are downloadable from Feem's web site "www.feem.it".

If you wish to receive hard copies you may choose from the payment options listed in the following table (minimum order: 10 papers)\*.

\*Orders for individual papers should clearly indicate the "Nota di Lavoro" number and can therefore be issued for published papers only.

All orders must be sent by fax to:

"Publications Office" - Fondazione Eni Enrico Mattei: Fax +39+2+52036946

### PAYMENT OPTIONS

How many papers?	What's the price?	How to pay?
10 or more*	US\$ 4.00 each Euro 5,00 each	By Credit card or Bank transfer
Annual subscription (approx. 100 papers/year)	US\$ 250.00 Euro 219,00	By Credit card or Bank transfer

\*Please fill out the Working Paper Subscription Form indicating your preferences (Periodic E-mail "Note di Lavoro" Bulletin, Annual subscription, Order for individual papers - minimum 10\*)!

#### WORKING PAPER SUBSCRIPTION FORM

Name:			
Affiliation(if applicable):			
Phone:	Fax:	E-mail:	
I wish to:			Amount due:
□ receive the Periodic E-mail	Working Papers Bulle	ətin	
D place a full annual subscrip	otion for 2002 (US\$ 250	).00/Euro 219,00)	
order noindividual pape	ers (minimum 10 paper	rs at US\$ 4.00/Euro 5,00 each)*	
Total			
I will pay by:			
□ VISA □ American Express	Card No	Expiration Date: _	·····
Signature:			
		ne Eni Enrico Mattei - account n	
Copy of the bank transfer sh	nould be faxed along	edito Artigiano - Corso Magenta with the order.	a 59, 20123 Milano, Italy.
Please return this duly com	pleted form to:		

"Publications Office" - Fondazione Eni Enrico Mattei - Corso Magenta, 63 - 20123 Milano, Italy